UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): January 31, 2023

Smith & Wesson Brands, Inc.

(Exact Name of Registrant as Specified in Charter)

Nevada (State or other jurisdiction of incorporation) 001-31552 (Commission File Number) 87-0543688 (IRS Employer Identification No.)

2100 Roosevelt Avenue Springfield, Massachusetts 01104 (Address of principal executive offices) (Zip Code)

(800) 331-0852

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

	Trading	Name of each exchange
Title of each class	Symbol(s)	on which registered
Common Stock, Par Value \$0.001 per Share	SWBI	Nasdaq Global Select Market

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (17 §CFR 230.405) or Rule 12b-2 of the Securities Exchange Act of 1934 (17 CFR §240.12b-2).

Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01. Entry into a Material Definitive Agreement.

Amended and Restated Guaranty.

On October 26, 2017, Smith & Wesson Brands, Inc. (formerly known as American Outdoor Brands Corporation) ("<u>SWBI</u>") entered into a guaranty (the "<u>Guaranty</u>) in favor of Ryan Boone County, LLC ("<u>Landlord</u>") related to a lease agreement, dated October 26, 2017, between Smith & Wesson Sales Company (formerly Smith & Wesson Corp.) ("<u>SWSC</u>") and Landlord concerning certain real property located in Boone County, Missouri, as subsequently amended on October 25, 2018 and January 31, 2019 (together, the "<u>Lease</u>"). On January 31, 2023, SWBI and SWSC entered into an amended and restated guaranty (the "<u>A&R Guaranty</u>") in favor of RCS – S&W Facility, LLC, as successor in interest to the Landlord, pursuant to which SWSC has been added as a guarantor.

The terms of the A&R Guaranty are substantially similar to the terms of the Guaranty.

The foregoing is a summary only and does not purport to be a complete description of all of the terms, provisions, covenants, and agreements contained in the A&R Guaranty, and is subject to and qualified in its entirety by reference to the full text of the A&R Guaranty, which is attached as Exhibit 10.134 and incorporated by reference herein.

Assignment and Assumption Agreement.

On January 31, 2023, SWSC and American Outdoor Brands, Inc. ("<u>AOB</u>") entered into an assignment and assumption of lease agreement (the "<u>Assignment Agreement</u>"), pursuant to which SWSC will assign, and AOB will assume, all of SWSC's rights, entitlement and obligations in, to and under the Lease, in each case effective on January 1, 2024 (the "<u>Assignment Date</u>"). The Assignment Agreement is subject to a number of conditions precedent, including, but not limited to, (i) delivery by SWSC to AOB and Landlord of executed copies of certain incentive assignment documents, including those related to a real property incentive agreement and a personal property incentive agreement; (ii) delivery by SWSC and AOB of written notice to Boone County, Missouri and the Missouri Development Finance Board; (iii) approval by Boone County, Missouri of the assignment prior to the Assignment Date; and (iv) receipt by SWSC of the Landlord Release (as defined therein). If any of the conditions precedent are not satisfied on or before the Assignment Date, each of SWSC and AOB may terminate the Assignment Agreement.

The foregoing is a summary only and does not purport to be a complete description of all of the terms, provisions, covenants, and agreements contained in the Assignment Agreement, and is subject to and qualified in its entirety by reference to the full text of the Assignment Agreement, which is attached as Exhibit 10.135 and incorporated by reference herein.

Design-Build Agreement.

On February 2, 2023, we entered into a design-build agreement (the "<u>Construction Contract</u>") with The Christman Company ("<u>Christman</u>") related to the construction of our new distribution facility and corporate office headquarters in Maryville, Tennessee, which we announced on September 30, 2021. The Construction Contract, which has an effective date of September 13, 2021, incorporates the arrangements under which we and Christman have been proceeding.

Pursuant to the Construction Contract, Christman is obligated to deliver certain services, including, among others, design phase services and construction phase services, and we are obligated to pay Christman for services performed. The parties to the Construction Contract have jointly agreed that Christman will perform, and complete, the Work (as defined therein) on a cost plus basis for a guaranteed maximum price of \$114,533,853, including contingencies. Due to the impact of inflation and supply chain issues, we continue to believe that we will spend more than \$120 million on the project in aggregate.

The Construction Contract includes terms that are customary for contracts of this type, including with respect to indemnification and insurance. The Construction Contract lists certain contract milestones and guaranteed completion dates, and we will be entitled to liquidated damages under certain circumstances. Each party to the Construction Contract is entitled to terminate the Construction Contract under certain circumstances.

The foregoing is a summary only and does not purport to be a complete description of all of the terms, provisions, covenants, and agreements contained in the Construction Contract, and is subject to and qualified in its entirety by reference to the full text of the Construction Contract, which is attached hereto as Exhibit 10.136 and incorporated by reference herein.

Safe Harbor Statement

Certain statements contained in this report may be deemed to be forward-looking statements under federal securities laws, and we intend that such forward-looking statements be subject to the safe harbor created thereby. Such forward-looking statements include, among others, that we continue to believe that we will spend more than \$120 million on the construction project in Tennessee in aggregate. We caution that these statements are qualified by important risks, uncertainties, and other factors that could cause actual results to differ materially from those reflected by such forward-looking statements. Such factors include, among others, economic, social, political, legislative, and regulatory factors; the impact of lawsuits; the state of the U.S. economy in general and the firearm industry in particular; general economic conditions and consumer spending patterns; the impact of inflation and supply chain disruptions; our ability to effectively manage and execute the planned relocation of our headquarters and certain of our operations to Tennessee; and other risks detailed from time to time in our reports filed with the Securities and Exchange Commission, including our Annual Report on Form 10-K for the fiscal year ended April 30, 2022.

Item 9.01. Financial Statements and Exhibits.

(d) Exhibits.

Exhibit No.	
10.134	Amended and Restated Guaranty, dated January 31, 2023.
10.135	Assignment and Assumption of Lease Agreement, dated January 31, 2023, between Smith & Wesson Sales Company and American Outdoor Brands, Inc.
10.136*	Standard Design-Build Agreement and General Conditions, dated February 2, 2023, between Smith & Wesson Brands, Inc. and The Christman Company_(portions of the exhibit have been omitted)

104

Cover Page Interactive Data File (embedded within the Inline XBRL document)

* Certain schedules and exhibits have bene omitted from this filing pursuant to Instruction 4 to Item 1.01 of Form 8-K. We agree to furnish supplementally a copy of any omitted schedule or exhibit to the Securities and Exchange Commission upon request.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: February 3, 2023

SMITH & WESSON BRANDS, INC.

By: /s/ Deana L. McPherson

Deana L. McPherson Executive Vice President, Chief Financial Officer, Treasurer, and Assistant Secretary

AMENDED AND RESTATED GUARANTY

THIS AMENDED AND RESTATED GUARANTY (this "Guaranty") is entered into by the undersigned (individually, collectively, jointly and severally, the "Guarantor") this 31st day of January, 2023 (the "Effective Date").

RECITALS

WHEREAS, SMITH & WESSON SALES COMPANY (formerly known as Smith & Wesson Corp.), a Delaware corporation ("S&W Tenant") and RCS – S&W FACILITY, LLC ("Landlord"), successor by assignment to Ryan Boone County, LLC, are parties to that certain Lease Agreement dated October 26, 2017, as amended by that certain First Amendment to Lease Agreement dated October 25, 2018, and as further amended by that certain Second Amendment to Lease Agreement dated January 31, 2019 (as amended, the "Lease");

WHEREAS, Smith & Wesson Brands, Inc., a Nevada corporation (the "Guarantor") (f/k/a American Outdoor Brands Corporation), executed that certain Guaranty dated October 26, 2017 in favor of Landlord (the "Original Guaranty");

WHEREAS, S&W Tenant desires to assign to AMERICAN OUTDOOR BRANDS, INC., a Delaware corporation ("AOB Tenant"), and AOB Tenant desires to assume, all of S&W Tenant's rights, entitlement, and obligations in, to and under the Lease (the "Lease Assignment");

WHEREAS, Landlord has agreed to consent to the Lease Assignment on the condition that the undersigned Guarantor execute this Guaranty in favor of Landlord.

AGREEMENT

NOW, THEREFORE, in consideration of the Lease, Lease Assignment, and matters described in the foregoing recitals, which recitals are incorporated herein and made a part hereof, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the undersigned agrees as follows for the benefit of Landlord and its successors, endorsees, transferees, participants and assigns as follows:

1. For purposes hereof, including, but not limited to, the foregoing recitals, all capitalized terms used, but not defined, in this Guaranty shall have the meaning ascribed to them in the Lease.

2. For purposes of defining "**Tenant**" hereunder, prior to the Assignment Date (as defined in the Lease Assignment), any reference to "Tenant" hereunder shall be deemed to refer to S&W Tenant, and on and following the Assignment Date (as defined in the Lease Assignment), any reference to "Tenant" hereunder shall be deemed to refer to AOB Tenant.

3. Subject to the other terms hereof, Guarantor hereby individually, collectively, jointly and severally: (i) guarantees to the Landlord, its successors and assigns, the full and prompt payment of all amounts due from Tenant to Landlord under and pursuant to said Lease during the Term, and the full performance and observance of all the covenants, terms, conditions and agreements therein provided to be performed and observed by the Tenant, its successors and assigns, under the Lease during the Term; and (ii) agrees that if default shall at any time be made by the Tenant, its successors and assigns, in the payment or in the performance of any of the terms, covenants, provisions or conditions contained in said Lease during the Term, then Guarantor will forthwith make such payments due to the Landlord, its successors and

assigns, under the Lease during the Term and will forthwith faithfully perform and fulfill all of such terms, covenants, conditions and provisions arising under the Lease during the Term, including, without limitation, the payment to the Landlord of all actual damages recoverable by Landlord under the Lease as a result of a Tenant Default, but excluding damages that are not recoverable under the Lease as a result of the limitations set forth therein. If Landlord notifies Tenant that it is in default under the Lease during the Term, Landlord shall simultaneously give written notice of such default to Guarantor, at the address set forth herein, and Guarantor shall have the right to remedy such default within the cure periods provided in the Lease; provided, such cure periods shall not commence until Landlord gives Guarantor written notice of such default. Landlord shall accept Guarantor's cure of any such default, and no Tenant Default shall arise therefrom in the event such default is timely cured.

4. This Guaranty is an absolute and unconditional Guaranty of payment and of performance. It shall be enforceable against the Guarantor, its successors and assigns, without the necessity for any suit or proceedings on the Landlord's part of any kind or nature whatsoever against the Tenant, its successors and assigns, and without the necessity of any notice of non-payment, non-performance or nonobservance or any notice of acceptance of this Guaranty or any other notice or demand to which the Guarantor might otherwise be entitled, all of which the Guarantor hereby expressly waives except as otherwise provided in the Lease or herein; and the Guarantor hereby expressly agrees that the validity of this Guaranty and the obligations of the Guarantor hereunder shall in no way be terminated, affected or impaired by reason of the assertion or the failure to assert by the Landlord against the Tenant, or the Tenant's successors and assigns, of any of the rights or remedies reserved to the Landlord pursuant to the provisions of said Lease.

5. Effect of Lease Assignment or Tenant Financial Status on Guaranty.

(a) If Tenant assigns the Lease and is fully released from its obligations and liabilities under the Lease by written agreement from Landlord or by the express terms of the Lease, then Guarantor shall be released from its obligations and liabilities under this Guaranty. If Tenant assigns the Lease and is partially released from its obligations and liabilities under the Lease (the "Released Obligations") by written agreement from Landlord or by the express terms of the Lease (including, but not limited to, Section 14.01 of the Lease), then Guarantor shall have no obligations or liabilities under this Guaranty with respect to the Released Obligations.

(b) For purposes hereof:

"Assumption Agreement" means a written agreement pursuant to which any Person assumes, in whole or in part, the obligations of Guarantor under this Guaranty for the benefit of Landlord.

"Net Worth" means, with respect to any Person, the difference between such Person's total assets and its total liabilities, excluding the amount of any negative goodwill and calculated in accordance with generally accepted accounting principles or other reputable accounting principles.

"Qualified Replacement Guarantor" means any Person that (i) has either a current long term unsubordinated debt rating of "BBB" or higher from Standard & Poor's Corporation ("Standard & Poor's") or a long term unsubordinated debt rating of "Baa" or higher from Moody's Investor Services ("Moody's) or a similar or higher rating published by a nationally recognized and reputable ratings agency or financial services firm or a similar or better creditworthiness evaluation prepared by a qualified financial services firm or national accounting firm; and (ii) has a Net Worth that equals or exceeds the Net Worth of Guarantor on the Effective Date. If Standard & Poor's or Moody's ceases to publish such debt ratings, Guarantor may substitute a similar debt rating scale published by a nationally recognized and reputable ratings agency or financial services firm for the same.

"Substitute Guaranty" means an agreement, in substantially the form of this Guaranty, pursuant to which any Person guarantees, in whole or in part, the satisfaction of Tenant's liabilities under the Lease for the benefit of Landlord.

If a Qualified Replacement Guarantor executes an Assumption Agreement or a Substitute Guaranty covering all of Guarantor's obligations under this Guaranty, then Guarantor shall be released from its obligations and liabilities under this Guaranty. If a Qualified Replacement Guarantor executes an Assumption Agreement under which it partially assumes Guarantor's obligations under this Guaranty or a Substitute Guaranty partially covering Guarantor's obligations under this Guaranty, then Guarantor shall be released from its obligations and liabilities under this Guaranty, to the extent, but only to the extent, the same are assumed by such Qualified Replacement Guarantor or covered by such Substitute Guaranty. Any release of Guarantor pursuant to this paragraph shall only be effective upon the delivery of the applicable Assumption Agreement or Substitute Guaranty to Landlord.

(c) Following the Assignment Date (as defined in the Lease Assignment), if Tenant satisfies the Financial Benchmarks, Guarantor shall be released from its obligations and liabilities under this Guaranty arising from Tenant's obligations and liabilities under the Lease accruing from and after the Financial Benchmark Date. For purposes hereof:

"Financial Benchmarks" means (i) Tenant either has a current long term unsubordinated debt rating of "BBB" or higher from Standard & Poor's, or a long term unsubordinated debt rating of "Baa" or higher from Moody's, or a similar or higher rating published by a nationally recognized and reputable ratings agency or financial services firm, or a similar or better creditworthiness evaluation prepared by a qualified financial services firm or national accounting firm and (ii) Tenant has a Net Worth that equals or exceeds \$390,645,000. If Standard & Poor's or Moody's ceases to publish such debt ratings, Landlord shall substitute a similar debt rating scale published by a nationally recognized and reputable ratings agency or financial services firm that is reasonably acceptable to Guarantor for the same.

"Financial Benchmark Date" means the date that Tenant satisfies the Financial Benchmarks.

6. All notices, consents, approvals and other communications (collectively, "Notices") that may be or are required to be given by either Guarantor or Landlord under this Guaranty shall be properly made only if in writing and sent to the address of Guarantor or Landlord, as applicable, set forth below, as the same is modified in accordance herewith, by hand delivery, U.S. Certified Mail (Return Receipt Requested) or nationally recognized overnight delivery service.

If to Guarantor:

Smith & Wesson Brands, Inc. c/o Smith & Wesson Corp. 2100 Roosevelt Avenue Springfield, MA 01104 Attn: General Counsel

If to Guarantor:	Smith & Wesson Sales Company c/o Smith & Wesson Corp. 2100 Roosevelt Avenue Springfield, MA 01104 Attn: General Counsel
If to Landlord:	RCS - S&W FACILITY, LLC c/o Real Capital Solutions, Inc. 371 Centennial Parkway, Suite 200 Louisville, CO 80027 Attn: Asset Manager
with Copy to:	Real Capital Solutions, Inc. 371 Centennial Parkway, Suite 200 Louisville, CO 80027 Attn: General Counsel

Either Guarantor or Landlord may change its address for Notices by giving a ten (10) day prior written notice to the other party in accordance with this provision. Notices shall be deemed received on the date of actual delivery; provided if either Guarantor or Landlord refuses to accept the delivery of any Notice, such notice shall be deemed to have been actually delivered on the date of such refusal.

7. Continuation.

(a) The liability of the Guarantor hereunder shall continue notwithstanding the incapacity, death, disability, merger, bankruptcy, insolvency, liquidation or dissolution of the Tenant, any Guarantor, any guarantor under any other instrument or any other person or entity primarily or secondarily liable for any amount or obligation under the Lease. The failure by the Landlord to file or enforce a claim against any estate (whether in administration, bankruptcy or other proceeding) of the Tenant, any Guarantor, any guarantor under any other instrument or of any other person or entity primarily or secondarily liable for any amount or obligation under the Lease, shall not affect the liability of the Guarantor hereunder.

(b) Should the status of the Tenant or Guarantor change, this Guaranty shall continue and also cover the Tenant or Guarantor under the new status or legal identity according to the terms hereof.

8. This Guaranty will be interpreted by and construed under the laws of the State of Missouri. Guarantor: (i) agrees that any suit, action or other legal proceeding arising out of or relating to this instrument shall be brought in a court of record in the State of Missouri or in the Courts of the United States of America having jurisdiction over the State of Missouri, (ii) consents to the jurisdiction of each such court in any such suit, action or proceeding and (iii) waives any objection which Guarantor may have to the laying of venue of any suit, action or proceeding in any of such courts and any claim that any such suit, action or proceeding has been brought in an inconvenient forum.

9. In the event any one or more of the provisions contained in this instrument or any application thereof shall be invalid, illegal, or unenforceable in any respect, the validity, legality or enforceability of the remaining provisions of this instrument and any other application thereof shall not in any way be affected or impaired.

10. Except as otherwise expressly provided herein, this Guaranty shall be a continuing Guaranty, and the liability of the Guarantor hereunder shall in no way be affected, modified or diminished by reason of the Lease Assignment or any future assignment of the Lease, any extension of the Lease pursuant to the terms thereof, or by reason of any waiver of any of the terms, covenants, conditions or provisions of said Lease by Landlord, or by reason of any extension of time that may be granted by the Landlord to the Tenant, its successors or assigns.

11. The Guaranty shall be binding upon and inure to the benefit of the successors and assigns of Guarantor and Tenant. Any references to Guarantor, Tenant and Landlord shall be deemed to mean and include the party and its respective successors and assigns. Guarantor shall not assign this Guaranty without the express written consent of the Landlord. Landlord shall have the right to assign and transfer this Guaranty to any Person to whom it assigns its rights under the Lease. Landlord's successors and assigns shall have the rights, elections, remedies, privileges, discretions and powers granted under this Guaranty to Landlord and the right to rely upon this Guaranty in the same manner and with the same force and effect as if they were specifically named as Landlord herein.

12. This Guaranty constitutes the entire agreement and understanding of Guarantor and Landlord with respect to the subject matter hereof and supersedes all prior agreements, understandings, letters, negotiations and discussions, whether oral or written. This Guaranty may be amended only by a written instrument executed by Guarantor and Landlord.

13. No waiver of any provision of this Guaranty shall be deemed to have been made unless expressed in writing and signed by the party charged therewith. The failure of Landlord to insist in any one or more instances upon the strict performance of any of the provisions of this Guaranty or to take advantage of any of its rights under this Guaranty shall not be construed as a waiver of any such provisions or the relinquishment of any such rights, rather the same shall continue and remain in full force and effect. No delay or omission in the exercise of any remedy accruing upon the breach of this Guaranty shall not be deemed a waiver of such breach. The waiver by Landlord of any breach of this Guaranty shall not be deemed a waiver of any other breach of the same or any other provision hereof.

14. Whenever the context may require, any pronoun used in this Guaranty shall include the masculine, feminine and neuter forms. All references to articles, sections and paragraphs of this Guaranty, unless the context shall indicate otherwise. The terms "hereof", "hereunder" and similar expressions refer to this Guaranty as a whole and not to any particular article, section or paragraph contained herein. The titles of the articles, sections and paragraphs of this Guaranty are for convenience only and shall not affect the meaning of any provision hereof. Guarantor and Tenant have agreed to the particular language of this Guaranty, and any question regarding the meaning of this Guaranty shall not be resolved by a rule providing for interpretation against the party who caused the uncertainty to exist or against the draftsman.

15. In the event any legal proceeding is commenced related to this Guaranty, the prevailing party in such proceeding shall be entitled to recover its reasonable attorneys' fees, court costs and litigation expenses from the non-prevailing party therein.

16. GUARANTOR HEREBY EXPRESSLY WAIVES ITS RIGHT TO A TRIAL BY JURY OF ANY CLAIM ARISING UNDER THIS GUARANTY, WHETHER NOW EXISTING OR HEREAFTER ARISING. LANDLORD MAY FILE AN ORIGINAL OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE FOREGOING WAIVER.

17. Guarantor acknowledges that Landlord has made no oral statements to Guarantor that could be construed as a waiver of Landlord's right to enforce this Guaranty by all available legal means.

[Signature to follow]

IN WITNESS WHEREOF, the parties hereto have entered into this Guaranty on the date first above written.

GUARANTOR:

SMITH & WESSON BRANDS, INC., a Nevada corporation

By: /s/ Mark Smith

Name:	Mark P. Smith
Title:	President

SMITH & WESSON SALES COMPANY, a Delaware corporation

By: /s/ Mark Smith

Name: Mark P. Smith Title: President

[Signature Page to Guaranty]

ASSIGNMENT AND ASSUMPTION OF LEASE AGREEMENT

THIS ASSIGNMENT AND ASSUMPTION OF LEASE AGREEMENT (this "Agreement") is made and entered into effective as of January 31, 2023, by and between SMITH & WESSON SALES COMPANY (formerly known as Smith & Wesson Corp.), a Delaware corporation ("Assignor"), and AMERICAN OUTDOOR BRANDS, INC., a Delaware corporation ("Assignee").

RECITALS:

WHEREAS, Assignor and RCS – S&W FACILITY, LLC ("Landlord"), successor by assignment to Ryan Boone County, LLC, are parties to that certain Lease Agreement dated October 26, 2017, as amended by that certain First Amendment to Lease Agreement dated October 25, 2018, and as further amended by that certain Second Amendment to Lease Agreement dated January 31, 2019 (as amended, the "Lease"), pursuant to which Assignor leases from Landlord certain improved real property described therein located in Boone County, Missouri, which property includes an office and warehouse building and other improvements (collectively, the "MO Property"); and

WHEREAS, Assignor and Assignee entered into that certain Sublease Agreement dated August 24, 2020, as amended by that certain First Amendment to Sublease Agreement dated as of July 16, 2022 (as amended, the "Existing Sublease"), pursuant to which Assignor subleased to Assignee, and Assignee subleased from Assignor, certain office space and warehouse space within said building; and

WHEREAS, Smith & Wesson Brands, Inc., a Nevada corporation (the "Guarantor") (f/k/a American Outdoor Brands Corporation) executed that certain Guaranty dated October 24, 2017 in favor of Landlord (the "Guaranty"); and

WHEREAS, upon the Assignment Date (as defined below), Assignor will assign to Assignee, and Assignee will assume, all of Assignor's rights, entitlement, and obligations in, to and under the Lease, as more particularly hereinafter set forth; and

WHEREAS, Assignor and Assignee desire to enter into this Agreement for the purpose of (1) evidencing said assignment and assumption of Lease, and (2) memorializing certain other agreements and understandings between them with respect thereto, as more particularly hereinafter set forth; and

WHEREAS, capitalized terms used but not defined herein shall have the meanings set forth in the Lease.

NOW, THEREFORE, for the covenants contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, Assignor and Assignee agree as follows:

ARTICLE I CONDITIONS PRECEDENT

1.01 <u>Conditions to Assignment and Assumption of Lease</u>. This Agreement shall be binding and enforceable against Assignor and Assignee upon the full execution and delivery hereof; provided, however, the provisions of Article II of this Agreement (including, without limitation, the assignment and assumption of the Lease) shall not be effective until January 1, 2024 (the "Assignment Date"), but only if, as of the Assignment Date, the following conditions (collectively, the "Conditions Precedent") are satisfied:

(i) Landlord has executed and delivered to Assignor and Assignee the Landlord Consent set forth below, following Assignor's and Assignee's signature page;

(ii) Assignor shall have received and provided executed copies to Assignee of the following documents (collectively, the "Incentive Assignment Documents"):

(A) an Assignment and Assumption Agreement among Assignor, Assignee, Landlord and Boone County, Missouri (the "**County**") related to the real property incentives for the MO Property (the "**Real Property Assignment Agreement**"), which provides for the assignment from Assignor to Assignee of Assignor's interest in that certain Performance Agreement dated as of December 1, 2017 between the County, Landlord, and Assignor and any related documents for which County consent is required (as amended, collectively, the "**Real Property Incentive Agreements**"), and pursuant to which the County consents to the Real Property Assignment Agreement and releases Assignor from all obligations under the Real Property Incentive Agreements that arise on or after the Assignment Date of the Real Property Assignment Agreement; and

(B) an Assignment and Assumption Agreement among Assignor, Assignee and the County related to certain personal property located at the MO Property (the "**Personal Property Assignment Agreement**"), which provides for the assignment from Assignor to Assignee of Assignor's interest in that certain Lease Agreement dated as of April 1, 2018 between the County and Assignor, that certain Bond Purchase Agreement dated as of April 1, 2018 between the County and Assignor and certain related documents (as amended, collectively, the "**Personal Property Incentive Agreements**"), and pursuant to which the County consents to the Personal Property Assignment and releases Assignor from all obligations under the Personal Property Incentive Agreements that arise on or after the effective date of the Personal Property Assignment Agreement; and

(C) an Assignment of Bond from Assignor to Assignee and an Assignee Representation Letter from Assignee related to the bond issued in connection with the Personal Property Incentive Agreements; and

(iii) Assignor and Assignee shall have provided written notice to the County and the Board, as applicable, pursuant to the Incentive Assignment Documents for the effective date to occur thereunder on the Assignment Date;

(iv) Landlord shall have executed and delivered to Assignor and Assignee an estoppel certificate in accordance with <u>Section 26.03</u> of the Lease, and without reference to any existing default or existing circumstances which, with the giving of notice, the passage of time, or both, would constitute a default under the Lease (other than defaults arising from the actions of Assignee or the employees, agents, contractors, customers, vendors, suppliers, visitors, invitees, licensees or subtenants of Assignee);

(v) Assignee, or an entity which is controlling and under common control with Assignee, shall have and shall maintain a long-term debt rating in any of the top three long term debt rating categories (i.e. AAA, AA, A, or their equivalent) by a nationally recognized rating agency and shall provide proof of the same to Assignor and Landlord on or before the Assignment Date ("Assignee Debt Rating Evidence"), or, alternatively, the County shall have approved to the assignment of the Lease from Assignor to Assignee in writing on or before the Assignment Date in accordance with Section 13.1(c) of the Prime Lease (as such term is defined in the Lease), which consent shall be in the form attached hereto as Exhibit A or such other form reasonably acceptable to Landlord (the "County Sublease Consent"); and

(vi) Assignor shall have received a written release from Landlord (in form and substance reasonably acceptable to Assignor and Landlord) pursuant to which Landlord releases Assignor from all obligations and liabilities under the Lease from and after the Assignment Date (the "Landlord Release"), provided that on or before the Assignment Date, Assignor and Guarantor shall provide to Landlord an amended and restated guaranty (in the form attached hereto as Exhibit B), pursuant to which Guarantor shall affirm, amend, and restate the Guaranty and Assignor and Guarantor shall jointly and severally guaranty Assignee's obligations to Landlord under the Lease terminating on the last day of the Term (as such term is defined in the Lease) (the "Amended and Restated Guaranty").

Commencing promptly upon the execution and delivery of this Agreement, Assignor shall endeavor diligently and in good faith to attain satisfaction of all the Conditions Precedent. Promptly upon the satisfaction of all Conditions Precedent, Assignor shall notify Assignee and Landlord, in writing, that the Conditions Precedent are satisfied. If any of the Conditions Precedent are not satisfied on or before January 1, 2024 (the "**Conditions Precedent Deadline**"), each of Assignor and Assignee may terminate this Agreement any time thereafter, but prior to the date that the notice of satisfaction of all Conditions Precedent is delivered pursuant to this paragraph (such termination notice to be concurrently sent to Landlord). The termination of this Agreement pursuant to this Section 1.01, if any, shall not affect the Existing Sublease, which shall remain in full force and effect if this Agreement is terminated, except as set forth in the Existing Sublease.

1.02 <u>Build Missouri Assignment or Credits</u>. Commencing promptly upon the execution and delivery of this Agreement, Assignor shall endeavor diligently and in good faith to obtain a Build Missouri Transaction Amendment among Missouri Development Finance Board (the "**Board**"), Assignor and Assignee (the "**Build Missouri Assignment Agreement**"), which provides for the assignment from Assignor to Assignee of Assignor's interest in that certain Amended and Restated Build Missouri Program Agreement dated as of August 20, 2020 among Assignor, Assignee and the Board and certain related documents (as amended, collectively, the "**Build Missouri Incentive Agreements**") and pursuant to which the Real Property Incentive Agreements and the Personal Property Incentive Agreements, the "**Incentive Agreements**") and pursuant to which the Board consents to the Build Missouri Assignment Agreement. If Assignor is unable to obtain the fully executed Build Missouri Assignment Agreement, with the Board's written consent and release of Assignor, prior to the satisfaction of the Conditions Precedent, such failure shall not be a default under this Agreement or the failure of a Condition Precedent; rather, Assignor shall execute and deliver to Assignee a written assignment in a form reasonably acceptable to Assignee, pursuant to which Assignor assigns, transfers and sets over to Assignee all of Assignor's right and interest in any and all payments and tax credits and other credits and benefits (if any) which are required to be provided to Assignor under the Build Missouri Assignment Agreement to Landlord.

1.03 <u>Cooperation</u>. Assignee agrees to cooperate, assist and join in Assignor's efforts to satisfy the Conditions Precedent. Without limiting the generality of the foregoing, (i) Assignee agrees to furnish Assignor with all information pertaining to Assignee and its operations at the MO Property that is reasonably requested in writing by Landlord or any of the other parties to the Incentive Agreements, including, without limitation, financial statements and information related to Assignee's employees (including, without limitation, number of employees, employment duties, and compensation); (ii) Assignee agrees to assume all of Assignor's obligations and liabilities under the Incentive Agreements; and (iii) Assignee agrees to protect, hold harmless, defend and indemnify Assignor from and against any and all claims, judgments, damages, liabilities, costs and expenses, including, without limitation, reasonable attorneys' fees and costs, accruing or relating to (a) Assignee 's obligations under the Incentive Agreements and which first arise on or after the Assignment Date, and (b) any failure by Assignee to comply with clause (i) of this section.

ARTICLE II ASSIGNMENT AND ASSUMPTION

The terms of this Article II shall take effect on the Assignment Date. Nothing herein shall be deemed to release either Assignor or Assignee from any obligation or liability arising under the Existing Sublease prior to the Assignment Date or as a result of any matter occurring prior the Assignment Date.

2.01 <u>Assignment</u>. On and effective as of the Assignment Date, Assignor hereby conveys, transfers and assigns to Assignee all of its right, title and interest in, to and under the Lease.

2.02 <u>Assumption</u>. On and effective as of the Assignment Date, Assignee hereby accepts the foregoing assignment and hereby assumes and agrees to perform and observe all of the covenants and agreements of "Tenant" under the Lease first arising from and after the Assignment Date.

2.03 <u>No Extension Options; Restrictions on Amending Lease</u>. Notwithstanding anything to the contrary herein or in the Lease, Assignee hereby agrees that Assignee shall not exercise any Extension Options under the Lease. Furthermore, so long as the Amended and Restated Guaranty is in effect, Assignee shall not amend or otherwise modify the Lease without Assignor's prior written consent, which consent may be granted or withheld by Assignor in its sole and absolute discretion.

2.04 <u>Proration</u>. All rent and other sums due and payable under the Lease shall be prorated between Assignor and Assignee as of the Assignment Date, with Assignor responsible for all sums due prior to the Assignment Date and Assignee responsible for all sums due from and after the Assignment Date.

2.05 <u>Indemnity by Assignor</u>. Assignor agrees to protect, hold harmless, defend and indemnify Assignee from and against any and all claims, judgments, damages, liabilities, costs and expenses, including, without limitation, reasonable attorneys' fees and costs, accruing or relating to Assignor's obligations under the Lease prior to the Assignment Date.

2.06 <u>Indemnity by Assignee</u>. Assignee agrees to protect, hold harmless, defend and indemnify Assignor from and against any and all claims, judgments, damages, liabilities, costs and expenses, including, without limitation, reasonable attorneys' fees and costs, accruing or relating to Assignee's obligations under the Lease on or after the Assignment Date.

2.07 <u>Payment for Personal Property</u>. On the Assignment Date, in connection with the assignment of Assignor's interest to Assignee of the personal property under the Personal Property Assignment Agreement (the "**Personal Property**"), Assignee shall pay to Assignor the net book value of the Personal Property as of the Assignment Date.

ARTICLE III ASSIGNEE INFORMATION

3.01 <u>Notification</u>. Upon request, Assignee shall provide Assignor and Landlord, in writing, a statement as to whether Assignee has met the following: (i) a current long term unsubordinated debt rating of "BBB" or higher from Standard & Poor's, or a long term unsubordinated debt rating of "Baa" or higher from Moody's, or a similar or higher rating published by a nationally recognized and reputable ratings agency or financial services firm, or a similar or better creditworthiness evaluation prepared by a qualified

financial services firm or national accounting firm (and for avoidance of doubt, nothing in this Agreement shall require Assignee to obtain any debt rating); and (ii) a Net Worth that equals or exceeds \$390,645,000. Such statements shall be furnished to Assignor and Landlord within five (5) Business Days after Assignor requests the same; provided that such request may only be made by Assignor after ninety (90) days of the end of Assignee's fiscal year; provided further that such request may also be made by Assignor after fifty (50) days of the end of Assignee's fiscal quarters if at any time Assignee reports (in response to a request by Assignor pursuant to this sentence) a Net Worth equal to or exceeding \$300,000,000.

3.02 <u>Cooperation</u>. In the event Assignee or any of its Affiliates satisfies the requirements of subparagraphs (i) and (ii) of Section 3.01 above, Assignor and Assignee shall (at no out-of-pocket cost to Assignee) endeavor, in good faith and with reasonable diligence, to request the Landlord to confirm, in writing, that Assignor has been partially released from its obligations and liabilities accruing from and after the Benchmark Date (as defined in the Amended and Restated Guaranty) under the Amended and Restated Guaranty in accordance with Section 5(c) of the Amended and Restated Guaranty.

ARTICLE IV MISCELLANEOUS

4.01 <u>Notices</u>. All notices, consents, approvals and other communications (collectively, "Notices") that may be or are required to be given by either Assignor or Assignee under this Agreement shall be properly made only if in writing and sent to the address of Assignor or Assignee, as applicable, set forth below, as the same is modified in accordance herewith, by hand delivery, or nationally recognized overnight delivery service.

If to Assignor:	Smith & Wesson Sales Company 2100 Roosevelt Avenue Springfield, MA 01104 Attn: President
with a copy to:	Smith & Wesson Brands, Inc. 2100 Roosevelt Avenue Springfield, MA 01104 Attn: General Counsel
If to Assignee:	American Outdoor Brands, Inc. 1800 N. Route Z Columbia, MO 65202 Attn: President
with a copy to:	American Outdoor Brands, Inc. 1800 N. Route Z Columbia, MO 65202 Attn: Chief Legal Officer

Either party may change its address for Notices by giving written notice to the other party in accordance with this provision. Notices shall be deemed received on the date of actual delivery; provided if either Assignor or Assignee refuses to accept the delivery of any Notice, such notice shall be deemed to have been actually delivered on the date of such refusal.

4.02 <u>Construction</u>. Unless the context indicates otherwise, (i) the terms "hereof", "hereunder," "herein" and similar expressions refer to this Agreement as a whole and not to any particular article, section or paragraph, (ii) the singular includes the plural and the masculine gender includes the feminine and neuter, and (iii) all references to articles, sections and subsections refer to the articles, sections and subsections of this Agreement. The titles of the articles, sections and subsections of this Agreement are for convenience only and shall not affect the meaning of any provision hereof. Assignor and Assignee have agreed to the particular language of this Agreement, and any question regarding the meaning of this Agreement shall not be resolved by a rule providing for interpretation against the party who caused the uncertainty to exist or against the draftsman. FOR PURPOSES OF THIS AGREEMENT, TIME SHALL BE CONSIDERED OF THE ESSENCE.

4.03 <u>Brokers</u>. Assignee and Assignor each (i) represents and warrants to the other that it has not dealt with any real estate broker, finder or listing agent in connection with this Agreement, and (ii) agrees to indemnify, defend and hold harmless the other from and against any claim for a commission, fee or other compensation made by a broker, finder or listing agent with whom it has dealt (or allegedly dealt). The provisions of this section shall survive the expiration or termination of this Agreement.

4.04 <u>Successors and Assigns</u>. This Agreement shall be binding on Assignor, Assignee and their respective successors and assigns. Assignee shall not assign this Agreement without Assignor's prior written consent, which consent may be granted or withheld by Assignor in its sole and absolute discretion. Any such assignment without Assignor's prior written consent shall be void.

4.05 Assignment and Subletting.

(a) <u>Assignment and Subletting</u>. Assignee shall not assign the Lease or sublet any portion of the MO Property, unless (i) Assignee has obtained Assignor's prior written consent (and Landlord's prior written consent if required under the Lease), which consent may be granted or withheld by Assignor in its sole and absolute discretion or (ii) Landlord shall have released Assignor from all of its obligations and liabilities under the Lease and any guaranty executed by Assignor related thereto, including, without limitation, Assignor's obligations under the Amended and Restated Guaranty. Assignee agrees to reimburse Assignor for all reasonable legal fees and other costs incurred by Assignor in connection with any permitted assignment of the Lease or any permitted subletting of the MO Property (or portion thereof) by Assignee, upon written demand. Assignee shall deliver to Assignor copies of all documents executed in connection with any permitted assignment of the Lease or any permitted subletting of the MO Property (or portion thereof) by Assignee, which documents must be in form and substance reasonably satisfactory to Assignor's consent to an assignment or subletting shall not be deemed a consent to any subsequent assignment or subletting. Nothing contained herein shall constitute an amendment to the Lease.

(b) <u>Transfer of Ownership Interests</u>. An assignment of this Agreement and the Lease by Assignee shall be deemed to have occurred if in a single transaction or in a series of transactions more than forty-nine percent (49%) of the ownership interests (whether stock, partnership interests, membership interests or other) in Assignee are, directly or indirectly, sold, transferred, conveyed or assigned; provided, the provisions of this clause (b) shall not apply to Assignee at any time when the stock of Assignee or its parent company is publicly traded on a recognized national securities exchange so long as Assignee is owned or controlled, directly or indirectly, by any Person who is engaged or whose Affiliate is engaged in the manufacture, assembly, sale, marketing or distribution of firearms, munitions and/or related accessories.

(c) <u>No Release</u>. Notwithstanding any assignment of this Agreement or the Lease or subletting of the MO Property, Assignee shall remain primarily responsible for the satisfaction of its obligations and liabilities hereunder and the satisfaction of the obligations and liabilities under the Lease, and such responsibility shall not be limited, affected, diminished or discharged by any event whatsoever, including, but not limited to, the compromise or settlement (with or without release) of any other person or entity liable for the payment of Assignee's liabilities and/or the performance of Assignee's obligations under this Agreement or the Lease, Assignor's failure to file suit against any assignee of Assignee (regardless of whether such assignee is becoming insolvent, is believed to be about to leave the state or any other circumstance), Assignor's failure to give Assignee notice of any default, the unenforceability of any provision of this Agreement or the Lease against an assignee of Assignee due to bankruptcy discharge or otherwise, Assignor's failure to insist upon the strict performance of the terms of this Agreement or the Lease, the extension, modification or amendment of this Agreement or the Lease or subletting of the MO Property (or portion thereof) not permitted under clause (a) of this section shall, at Assignor's option, be void.

4.06 Landlord's Conditions Precedent. Landlord's consent to this Agreement and the Landlord Release shall only be considered effective upon the following conditions being met on or prior to the Assignment Date, which include delivery to Landlord of the following items: (i) the fully executed Real Property Assignment Agreement; (ii) copies of all existing Personal Property Incentive Agreements; (iii) the fully executed Incentive Assignment Documents; (iv) the executed County Sublease Consent or Assignee Debt Rating Evidence; (v) the fully executed Amended and Restated Guaranty in the form attached hereto as Exhibit B; (vi) an Estoppel Certificate executed by Assignee in the form attached hereto as Exhibit C; (vii) a Subordination, Non-Disturbance and Attornment Agreement executed by Assignee in the form attached hereto as Exhibit D; and (viii) a copy of an agreement terminating the Existing Sublease. If the foregoing conditions precedent are not met on or before the Assignment Date Landlord's consent to this Agreement and the Landlord Release shall be deemed null and void.

4.07 <u>Relationship</u>. Nothing contained in this Agreement shall be deemed or construed as creating a partnership, joint venture, agency or other similar relationship between Assignor and Assignee.

4.08 <u>Severability</u>. If any provision of this Agreement is found by a court of competent jurisdiction to be illegal, invalid or unenforceable, the remainder of this Agreement will not be affected, and in lieu of each provision that is found to be illegal, invalid or unenforceable, a provision will be added as a part of this Agreement that is as similar to the illegal, invalid or unenforceable provision as may be possible and be legal, valid and enforceable.

4.09 Entire Agreement and Modification. This Agreement constitutes the entire agreement between the parties with respect to the subject matter hereof, and all prior negotiations and understandings shall be deemed incorporated herein. This Agreement may only be amended or modified by a written instrument signed by both Assignor and Assignee.

4.10 <u>No Waiver</u>. No waiver by Assigner or Assignee of any provision or breach of this Agreement shall be deemed to have been made unless the same is in writing, and no waiver of any provision or breach of this Agreement shall be deemed a waiver of any other provision or breach. Assigner's or Assignee's consent to or approval of any act shall not be deemed to render unnecessary the obtaining of Assigner's or Assignee's consent to or approval of any subsequent act.

4.11 <u>Submission</u>. The submission of this Agreement does not constitute an offer, and this document shall become effective and binding only upon the execution and delivery hereof by both Assignor and Assignee. Furthermore, copies of this Agreement that have not been executed and delivered by both Assignor and Assignee shall not serve as a memorandum or other writing evidencing an agreement between the parties.

4.12 <u>Attorney Fees</u>. In the event of any lawsuit between the parties arising from or relating to this Agreement, the prevailing party in such lawsuit shall be entitled to recover its reasonable costs, expenses and attorneys' fees from the non-prevailing party therein, including, but not limited to, court costs, professional fees and other litigation expenses through all appellate levels and in bankruptcy court. This section shall survive the expiration or termination of this Agreement.

4.13 Exhibits. Assignor and Assignee acknowledge and agree that all exhibits and addenda referenced in this Agreement are attached hereto and incorporated herein by reference.

4.14 Governing Law. This Agreement shall be governed by the laws of the State of Missouri.

4.15 <u>WAIVER OF TRIAL BY JURY</u>. EACH OF ASSIGNOR AND ASSIGNEE HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION (i) ARISING UNDER THIS AGREEMENT (OR ANY AGREEMENT FORMED PURSUANT TO THE TERMS HEREOF) OR (ii) IN ANY MANNER CONNECTED WITH OR RELATED OR INCIDENTAL TO THE MO PROPERTY, THE DEALINGS OF ASSIGNOR AND ASSIGNEE WITH RESPECT TO THIS AGREEMENT (OR ANY AGREEMENT ENTERED INTO PURSUANT TO THE TERMS HEREOF) OR ANY OTHER INSTRUMENT, DOCUMENT OR AGREEMENT EXECUTED OR DELIVERED IN CONNECTION WITH THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREIN, WHETHER NOW EXISTING OR HEREINAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT, TORT OR OTHERWISE. EITHER PARTY MAY FILE A COPY OF THIS SECTION WITH ANY COURT AS CONCLUSIVE EVIDENCE OF THE CONSENT OF EACH SUCH PARTY TO THE WAIVER OF ITS RIGHT TO TRIAL BY JURY.

4.16 <u>Assignor's Liability</u>. Notwithstanding anything to the contrary contained in this Agreement, the liability of Assignor to Assignee for the default by Assignor under the terms of this Agreement shall be limited to Assignee's actual damages, and no consequential, punitive or exemplary damages or lost profits shall be recoverable from Assignor. Assignor's shareholders, partners, members, officers and directors shall have no personal liability whatsoever under this Agreement.

4.17 <u>Counterparts and Signatures</u>. This Agreement may be executed in separate counterparts and it shall be fully executed when each party whose signature is required has signed at least one (1) counterpart even though no one (1) counterpart contains the signatures of all of the parties to this Amendment. Electronic signatures shall be valid and sufficient to bind any party to this Agreement. Signatures to this Agreement transmitted by facsimile, email or other electronic transmission (for example, through the use of a Portable Document Format or "PDF" file) shall be valid and effective to bind the party so signing. The exchange of copies of this Agreement and of signature pages by electronic mail or other means of electronic transmission (including, without limitation, pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) will constitute effective execution, delivery and performance of this Agreement as to the parties. Signatures of the parties transmitted by electronic mail or other means of electronic transmission (including, without limitation, pdf or any electronic signature complying with the U.S. federal ESIGN Act of 2000, e.g., www.docusign.com) will be deemed to be their original signatures for all purposes.

4.18 <u>Guarantor</u>. Guarantor hereby consents to this Agreement, ratifies and reaffirms the Guaranty and in the event of the consummation of this Agreement, the Amended and Restated Guaranty, as applicable.

[signature page follows]

IN WITNESS WHEREOF, the parties hereto have entered into this Agreement on the date first above written.

ASSIGNOR:

SMITH & WESSON SALES COMPANY

By: /s/ Mark Smith

Mark P. Smith President

ASSIGNEE:

AMERICAN OUTDOOR BRANDS, INC.

By: /s/ Brian Murphy

Brian D. Murphy President

GUARANTOR'S CONSENT:

Smith & Wesson Brands, Inc.

By: /s/ Mark Smith Mark P. Smith President

Dated as of January 31, 2023.

LANDLORD CONSENT:

By its execution below and subject to Landlord's Conditions Precedent set forth in Section 4.06 above, RCS – S&W FACILITY, LLC, as Landlord under the above-described Lease, hereby consents to this Agreement and to the assignment of the Lease from Assignor to Assignee pursuant to the terms and conditions set forth in this Agreement, and with respect thereto, this consent is provided in full satisfaction of the requirements of Section 14.01(b) and any other applicable provisions of the Lease requiring Landlord's consent or approval of the assignment of the Lease from Assignor to Assignee pursuant to this Agreement, but shall in no way constitute an amendment to the Lease.

Dated as of January 31, 2023.

RCS – S&W FACILITY, LLC, a Colorado limited liability company

By: /s/ Judy Lawson

Name: Judy Lawson Title: Manager

Exhibit A

CONSENT OF BOONE COUNTY, MISSOURI

This Consent by Boone County, Missouri ("County") is given in connection with that Assignment and Assumption of Lease Agreement dated effective as of ______, 202_ ("Assignment and Assumption Agreement"), by and between Smith & Wesson Sales Company (formerly known as Smith & Wesson Corp.), a Delaware corporation as Assignor ("Assignor") and American Outdoor Brands, Inc., a Delaware corporation as Assignee ("Assignee"), regarding that Lease Agreement by and between Smith & Wesson Sales Company (formerly known as Smith & Wesson Corp.) as Tenant and RCS – S&W Facility, LLC (successor to Ryan Boone County, LLC) as Landlord dated October 26, 2017, as amended by that certain First Amendment to Lease Agreement dated October 25, 2018, and as further amended by that certain Second Amendment to Lease Agreement dated January 31, 2019 (as amended, the "Lease"), pursuant to which Tenant leases from Landlord certain improved real property described therein located in Boone County, Missouri, which property includes an office and warehouse building and other improvements.

County hereby acknowledges and consents to the Assignment and Assumption Agreement and to the assignment of the Lease from Assignor to Assignee pursuant to the terms and conditions set forth in the Assignment and Assumption Agreement, and with respect thereto, this Consent is provided in full satisfaction of the requirements of Section 13.1(c) and any other applicable provisions of that Lease Agreement dated as of December 1, 2017 by and between Boone County, Missouri as Lessor and Ryan Boone County, LLC as Lessee which was thereafter assigned by Ryan Boone County, LLC to RCS – S&W Facility, LLC requiring the County's consent or approval of the assignment of the Lease from Assignor to Assignee pursuant to the Assignment and Assumption Agreement.

BOONE COUNTY, MISSOURI

[SEAL]

By:

Name: Daniel K. Atwill Title: Presiding Commissioner

ATTEST:

By: Name: Brianna L. Lennon Title: County Clerk

Exhibit B

AMENDED AND RESTATED GUARANTY

Exhibit C ESTOPPEL CERTIFICATE

r

Exhibit D SUBORDINATION, NON-DISTURBANCE AND ATTORNMENT AGREEMENT

E

CERTAIN IDENTIFIED INFORMATION HAS BEEN EXCLUDED BECAUSE IT IS NOT MATERIAL AND IS THE TYPE WE TREAT AS PRIVATE OR CONFIDENTIAL



ConsensusDocs[®] 410 STANDARD DESIGN-BUILD AGREEMENT AND GENERAL CONDITIONS BETWEEN OWNER AND DESIGN-BUILDER (Cost of the Work Plus a Fee with a GMP)

TABLE OF ARTICLES

1.	AGREEMENT	1
2.	GENERAL PROVISIONS	2
3.	DESIGN-BUILDER'S RESPONSIBILITIES	5
4.	OWNER'S RESPONSIBILITIES	20
5.	SUBCONTRACTS	21
6.	TIME	22
7.	PAYMENT	24
8.	COST OF THE WORK	26
9.	CHANGES IN THE WORK	27
10.	PAYMENT FOR CONSTRUCTION PHASE SERVICES	30
11.	INDEMNITY, INSURANCE, AND BONDS	34
12.	SUSPENSION, NOTICE TO CURE, AND TERMINATION	38
13.	DISPUTE MITIGATION AND RESOLUTION	40
14.	MISCELLANEOUS PROVISIONS	41
15.	CONTRACT DOCUMENTS	42

ARTICLE 1 AGREEMENT

Job Number: 221102

Account Code: N/A

This Agreement is effective as of the 13th day of September in the year 2021 ("Effective Date"), by and between Owner Smith & Wesson Brands, Inc., 2100 Roosevelt Ave., Springfield, MA 01104 and Design-Builder The Christman Company, 2400 Sutherland Ave., Knoxville, TN 37919

Tax identification number (TIN) FEIN #38-0419740 Contractor Licensing No., if applicable TN License No. 64704 Design Professional Licensing No. in the state of the Project TN License No. 1900

for services in connection with the following

PROJECT: "Project Eagle", Blount County, TN - New Distribution Facility & Corporate Office Headquarters



ConsensusDocs[®] **410 – Standard Design-Build Agreement and General Conditions Between Owner and Design-Builder (Cost of Work Plus Fee with GMP)** – [©] **2011, October Revised 2017.** THIS DOCUMENT MAY HAVE BEEN MODIFIED. The ConsensusDocs technology platform creates a redline comparison to the standard language which the purchaser of this contract is authorized to share for review purposes. Consultation with legal and insurance counsel are strongly encouraged. You may only make copies of finalized documents for distribution to parties in direct connection with this contract. Any other uses are strictly prohibited. CONTENT SECURE ID: D1625BE0-C8C7

ARTICLE 2 GENERAL PROVISIONS

2.1 TEAM RELATIONSHIP AND ETHICS Each Party agrees to act on the basis of good faith and fair dealing, and shall take all actions reasonably necessary to perform this Agreement in an economical and timely manner, including consideration of design modifications and alternative materials or equipment that will permit the Work to be constructed within the Guaranteed Maximum Price (GMP) and, if established, by the Dates of Substantial Completion and Final Completion, as well as the other Contract Milestones.

2.1.1 Neither Design-Builder nor any of its agents or employees shall act on behalf of or in the name of Owner.

2.2 ETHICS Each Party shall perform with integrity. Each shall: (a) avoid conflicts of interest; (b) promptly disclose to the other Party any conflicts of interest which may arise. Each Party warrants it has not and shall not pay or receive any contingent fees or gratuities to or from the other Party, including its agents, officers and employees, Design Professional, Subcontractors, Subsubcontractors, Suppliers, or Others, to secure preferential treatment.

2.3 DESIGN-PROFESSIONAL Design-Builder shall furnish architectural and engineering services ("Services") by Design-Builder's licensed employees or procure such services from a licensed, independent design professional/s retained by Design-Builder. The person or entity providing architectural and engineering services shall be referred to as Design-Professional. If Design-Professional is an independent design professional, the architectural and engineering services shall be procured pursuant to a separate agreement between Design-Builder and Design-Professional.

2.3.1 Design-Builder has and shall furnish or provide the Services necessary to design the Project in accordance with Owner's requirements, as outlined in Owner's Program and other relevant data defining the Project. The Owner has approved, the IFC drawings and Specifications identified in <u>Exhibit E to Exhibit 1 ("Current Drawings and Specifications"</u>), as consistent with and conforming to the Owner's Program. The Services shall be performed in accordance with the standard of professional skill and care required for a Project of similar size, scope, location, and complexity, during the time in which the Services are provided, as efficiently as is consistent with such professional skill and care and the orderly progress of the Work. ("Design Standard of Care"). Design-Builder shall have the right to reasonably rely upon manufacturer representations and specifications as to material, product, and equipment performance.

2.4 DEFINITIONS

2.4.1 "Agreement" means this ConsensusDocs 410 Standard Design-Build Agreement and General Conditions Between Owner and Design-Builder (Cost of the Work Plus a Fee with a GMP), as modified, amendments, exhibits, addenda, and attachments made part of this agreement upon its execution.

2.4.2 The following exhibits are a part of this Agreement:

Exhibit 1: GMP Proposal Exhibit E to Exhibit 1, Log of Current Drawings Exhibit G to Exhibit 1, Project Schedule Exhibit H to Exhibit 1, Schedule of Values

Exhibit 2: Key Personnel Exhibit 3-1: Conditional Interim Waiver Exhibit 3-2: Unconditional Interim Waiver Exhibit 3-3: Conditional Final Waiver Exhibit 3-4: Unconditional Final Waiver Exhibit 4: LNTP

2.4.3 "Business Day" means all Days, except weekends and official federal or state holidays where the Project is located.

2.4.4 A "Change Order" is a written order signed by the Parties after execution of this Agreement, authorizing a change in the scope of the Work, Guaranteed Maximum Price, or Contract Milestones, including substitutions proposed by Design-Builder and accepted by Owner.

2.4.5 "Construction Schedule" is the document prepared by Design-Builder, consistent with the Project Schedule and Contract Milestones, that specifies the dates on which Design-Builder plans to begin and complete various parts of the construction phase services Work, and the Project, including dates on which information and approvals are required from Owner.

2.4.6 The "Contract Documents" consist of those documents identified in §15.1.

2.4.7 The "Contract Milestones" are the guaranteed milestones, including Substantial Completion of the Work and Final Completion as defined in §6.2.

2.4.8 "Cost of the Work" means the costs and discounts specified in ARTICLE 8.

2.4.9 "Day" means calendar day.

2.4.10 "Date of Commencement" is as provided for in §6.1

2.4.11 "Design-Builder's Fee" means the compensation paid to Design-Builder for salaries and other mandatory or customary compensation of Design-Builder's employees at its principal and branch offices except employees listed in §8.2, general and administrative expenses of Design-Builder's principal and branch offices other than the field office, and Design-Builder's capital expenses, including interest on Design-Builder's capital employed for the Work, and profit.

2.4.12 "Defective Work" is any portion of the Work that does not conform to the requirements of the Contract Documents.

2.4.13 "Final Completion" occurs on the date when Design-Builder's obligations under this Agreement are complete and accepted by Owner and final payment becomes due and payable.

2.4.14 "Good Industry Practices" means those prudent and advisable practices, methods, techniques, acts and equipment generally followed or used by reputable professional design, engineering or construction contractors regularly involved in the design and construction of work of a similar size, scope and complexity as the Work in a similar geographic location ("Good Engineering and Construction Contractors") that, in the exercise of prudent, proper and professional judgment, in light of the facts known or that reasonably should have been known at the time a decision was made, would have been expected to accomplish the desired result in a manner consistent with reasonable ease, that would ordinarily be expected from Good Engineering and Construction Contractors

2.4.15 A Hazardous Material is any substance or material identified now or in the future as hazardous under any Laws or any other substance or material which may be considered hazardous or otherwise subject to statutory or regulatory requirements governing handling, disposal, or clean-up.

2.4.16 "Interim Directive" is any written order containing Work instructions that is signed by Owner after execution of this Agreement and prior to Substantial Completion of the Work directed by Owner pursuant to §9.3.

2.4.17 "Law" means a federal, state, or local law, ordinance, code, rule, or regulation applicable to the Work with which Design-Builder must comply that is enacted as of the Agreement date.

2.4.18 "Others" means Owner's other: (a) contractors/constructors/design-builders, (b) suppliers, (c) subcontractors, subsubcontractors, or suppliers of (a) and (b); and others employed directly or indirectly by (a), (b), or (c) or any by any of them or for whose acts any of them may be liable.

2.4.19 "Overhead" shall mean (a) payroll costs and other compensation of Design-Builder's employees in Design-Builder's principal and branch offices; (b) general and administrative expenses of Design-Builder's principal and branch offices including charges against Design-Builder for delinquent payments; and (c) Design-Builder's capital expenses, including interest on capital used for the Work.

2.4.20 "Owner" is the person or entity identified in ARTICLE 1, and includes Owner's Representative.

2.4.21 "Owner's Program" is an initial description of Owner's objectives that may include budget and time criteria, space requirements and relationships, flexibility and expandability requirements, special equipment and systems, and site requirements. In this case, that includes: Owner's request for the design and construction of a new headquarters with office space, a production and distribution and warehousing facility, an indoor range and PIM area to accommodate its owner provided equipment on a graded and secured site within the project budget/GMP as further refined and clarified in the discussions leading up to, and embodied in, the GMP Proposal.

2.4.22 The "Parties" are collectively Owner and Design-Builder.

2.4.23 The "Project," as identified in ARTICLE 1, is the building, facility, or other improvements for which Design-Builder is to perform the Work under this Agreement. It may also include improvements to be undertaken by Owner or Others.

2.4.24 "Project Schedule" is a schedule that shows the timing and sequencing of the design and construction required to meet the time criteria set forth in Owner's Program. The Project includes the Critical Milestones as well as the construction activities and is coordinated with design phase service activities.

2.4.25 A "Subcontractor" is a person or entity retained by Design-Builder as an independent contractor to provide the labor, materials, equipment, or services necessary to complete a portion of the Work. The term Subcontractor does not include Design-Professional or any separate contractor employed by Owner or any separate contractors.

2.4.26 "Substantial Completion" of the Work, or of a portion thereof, or of a designated Contract Milestone, occurs on the date when the Work is sufficiently complete in accordance with the Contract Documents so that Owner may occupy or utilize the Project, or a designated portion, for the use for which it is intended, without unapproved disruption. The issuance of a certificate of occupancy is not a prerequisite for Substantial Completion of the Work, so long as the certificate of occupancy cannot be obtained due to factors completely beyond Design-Builder's control. The date of Substantial Completion of each Contract Milestone shall be confirmed by a certificate of Substantial Completion signed by the Parties.

2.4.27 "Subsubcontractor" is a party or entity who has an agreement with a Subcontractor or another Subsubcontractor or Supplier to perform any portion of the Work or to supply material or equipment.

2.4.28 A "Supplier" is a person or entity retained by Design-Builder to provide material or equipment for the Work.

2.4.29 "Terrorism" means a violent act, or an act that is dangerous to human life, property, or infrastructure, that is committed by an individual or individuals and that appears to be part of an effort to coerce a civilian population or to influence the policy or affect the conduct of any government by coercion. Terrorism includes, but is not limited to, any act certified by the United States Secretary of Treasury as an act of terrorism pursuant to the Terrorism Risk Insurance Act, as amended.

2.4.30 The "Work" means the design phase services procured or furnished in accordance with §3.1, the GMP Proposal attached as **Exhibit 1**, the Construction Phase services provided in accordance with §3.3, and other services which are necessary to complete the Project in accordance with and reasonably inferable from the Contract Documents, including all work previously performed related to the Project under or preceding the February 2022 LNTP attached as **Exhibit 4** ("LNTP"). The Work may refer to the whole Project or only a part of the Project if work is also being performed by Owner or Others. "Work" specifically includes:

- (a) all work and services required in connection with the design, engineering, procurement, permitting, construction, construction management, temporary and permanent material, equipment and structures, inspection, expediting transportation, shipment, delivery, importation, installation, commissioning, testing and completion of the Work (including the start-up of its systems and components) in accordance with the Contract Documents;
- (b) the provision of all materials and equipment, machinery, tools, labor, chemicals, lubricants consumables, transportation, administration, oversight, incidentals and other services and items related to the foregoing in accordance with the Contract Documents;
- (c) the correction of defects and any actions required to comply with the warranty or Design-Builder's other post-completion obligations; and
- (d) any Additional Services, Changes, or Directives authorized by Owner;

provided that, "Work" shall not include any of Owner's obligations set forth or referenced in Article 4, the Owner's Program, or the GMP Proposal (**Exhibit 1**).

2.4.31 "Worksite" means the area of the Project location identified in ARTICLE 1 where the Work is to be performed.

ARTICLE 3 DESIGN-BUILDER'S RESPONSIBILITIES

Design-Builder shall be responsible for procuring or furnishing the design and for the construction of the Work consistent, and in conformance, with Owner's Program, and the requirements herein, as such Program may be modified by Owner during the course of the Work. Design-Builder shall use its diligent efforts to perform the Work in an expeditious manner, and warrants and guarantees schedules and the GMP set forth herein. Design-Builder represents that it is an independent contractor and that it is familiar with the type of Work required by this Agreement.

The intent of this Owner's Program is to prescribe the general requirements for the design, engineering, procurement, permitting, construction, and related services necessary for the safe and timely completion of the Work, which Design-Builder undertakes to perform in accordance with the terms and conditions of this Agreement.

Design-Builder shall perform or cause to be performed, on a timely design-build basis, for no more than the Guaranteed Maximum Price, the Work and provide Owner with a facility as described in, or reasonably inferable from the Contract Documents, capable of being operated and maintained in accordance with this Agreement and otherwise comply with Law, that will meet the specifications and other conditions of this Agreement.

Design-Builder shall do and shall provide or procure all things necessary for the safe and successful completion of the Work, including, but not limited to, as specified in Section 3.1 and Section 3.3. Without limiting the foregoing, Design-Builder shall:

- i. perform all site and locational, geotechnical, and hydrogeological investigation, design, specification, engineering and design and select materials, import materials, goods, processes, methods of work and technology;
- ii. project manage, coordinate, schedule, supervise and execute or cause to be executed all aspects of the Work, whether performed directly by Design-Builder or any Subcontractor;
- iii. be responsible, in liaison with Owner, for interfacing with, and coordinating the execution of the Work with the work of other contractors and of third parties (if any);
- iv. procure all materials, equipment or other goods or things required in connection with the Work;
- v. inspect, expedite and provide for quality assurance and quality control of (i) all such materials, equipment and other goods or things and (ii) the Work;
- vi. transport, store, protect, maintain and secure all such materials, equipment and other goods or things;
- vii. construct, erect, install, excavate, support, reinforce, modify, and do all permanent and temporary work required as part of the Work, including all site preparation and formation;
- viii. construct and maintain all infrastructure, roads, and bridges throughout Worksite;
- ix. review the permanent and temporary work previously carried out by others and intended to form part of the Work and advise Owner of any apparent errors. Notwithstanding anything to the contrary, Design Builder shall not assume any responsibility or liability for work performed by others, including the state, federal, or local entities, including but not limited to offsite utilities and public road improvement.;
- x. supply all labor and all supervisory, management, technical, administrative and other staff necessary for the execution of the Work;
- xi. correct any defects in the Work;
- xii. provide timely responses to requests for information made by Owner; and
- xiii. cooperate with the efforts of Owner and any Owner's Nominee in connection with this Agreement.

3.1 DESIGN PHASE SERVICES

The Agreement is a design-build agreement for the Work on a Cost Plus Basis with a Guaranteed Maximum Price and, in addition to its construction obligations, the Design-Builder shall be responsible for all design and engineering necessary for the completion of the Work as set forth in the Contract Documents. Design-Builder represents that it has thoroughly familiarized itself with the Project, the Worksite, the local conditions, and all design requirements for the Work (including the requirements set forth in Owner's Program). Design-Builder further agrees that it shall provide all engineering and design services necessary for the Work that meet the requirements set forth in the Agreement or otherwise required by the Design Standard of Care. All design or engineering work requiring certification shall be certified by professional engineers licensed and properly qualified to perform such certifications under applicable Laws. Any representation on LEED credit submittals is made solely for the purpose of addressing system credits and is not intended as a guarantee or warranty of certification, functionality, or performance.

3.1.1 Design-Builder has reviewed Owner's Program to ascertain the requirements of the Project and verified such requirements with Owner. Design-Builder has evaluated the site, including access, traffic, drainage, parking, building placement, and other considerations affecting the building, the environment, and energy use, as well as information regarding Laws and requirements.

3.1.2 OMITTED

3.1.3 OMITTED.

3.1.4 SCHEMATIC DESIGN DOCUMENTS Design-Builder has developed and Owner has approved Schematic Design Documents illustrating the Project's basic elements, scale, and their relationship to the Worksite at a conceptual level.

3.1.5 PLANNING PERMITS Design-Builder shall obtain and Owner shall pay for all planning permits necessary for the construction of the Project.

3.1.6 DESIGN DOCUMENTS Owner has approved as consistent with and conforming to the intent of Owner's Program, the Current Drawings and Specifications which further define the Project, including drawings and outline specifications fixing and describing the Project size and character as to site utilization, and other appropriate elements incorporating the structural, architectural, mechanical, and electrical systems. Owner's approval of, or failure to provide any comments to any of the Current Drawings and Specifications shall in no event be deemed as Owner's waiver of any of Design-Builder's obligations to perform the Work or Design-Builder's liability in accordance with the requirements of this Agreement.

3.1.7 FURTHER MODIFICATION OF CURRENT DRAWINGS Design-Builder shall identify in writing all material changes and deviations that it proposes from the Current Drawings. Owner may provide comments to any such material changes or deviations and, if Owner does so, it shall do so as soon as reasonably practicable after Owner's notification so as not to impact the progress of the Work. If modifications to the Current Drawings or Specifications are required, Design-Builder shall promptly submit to the Owner for review any required modifications. Design-Builder shall incorporate or address, in writing, Owner's comments to the Construction Documents to the reasonable satisfaction of Owner. Notwithstanding the foregoing, if Design-Builder believes that any comment from Owner constitutes a Change that is otherwise permissible and not prohibited by Article 9 or other portions of the Agreement, it shall comply with Article 9 before incorporating any such comment. If Design-Builder does not comply with Article 9, including providing written notice within ten (10) Days, before incorporating any comment of Owner, Design-Builder shall have waived its ability to request a Change under Article 9 in respect of such comment. Owner's approval of, or failure to provide any comments shall in no event be deemed as Owner's waiver of any of Design-Builder's obligations to perform the Work or Design-Builder's liability in accordance with the requirements of this Agreement.

3.1.8 OWNER PROVIDED INFORMATION Owner (and/or its consultants or representatives) has provided Design-Builder with (or has offered to make available for Design-Builder's review) certain information and documents related to the Project (all information and documents provided by Owner and/or its consultants or representatives is referred to as "Owner Provided Information"). Design-Builder shall be entitled to reasonably rely upon the accuracy of Owner Provided Information. Prior to executing the Agreement, Design-Builder represents that it has independently reviewed and evaluated the Worksite and analyzed the surrounding surface and subsurface areas and ground conditions, making Design-Builder's own independent evaluations.

3.1.9 OWNERSHIP OF DOCUMENTS

3.1.9.1 OWNERSHIP OF TANGIBLE DOCUMENTS Owner shall receive ownership of the property rights, except for copyrights, of all documents, drawings, specifications, electronic data, and information ("Documents") prepared, provided, or procured by Design-Builder, its Design-Professional, Subcontractors, or consultants and distributed to Owner for this Project, upon the making of final payment to Design-Builder or, in the event of termination under ARTICLE 12, upon payment for all sums due to Design-Builder pursuant to ARTICLE 12. Owner's acquisition of the copyright shall be subject to Owner's making of all payments required by this Agreement.

3.1.9.2 COPYRIGHT The Parties agree that Owner \Box shall/ \boxtimes shall not obtain ownership of the copyright of all Documents. Owner's acquisition of the copyright for all Documents shall be subject to the making of payments as required by the subsection above and the payment of the fee reflecting the agreed value of the copyright set forth below:

If the Parties have not made a selection to transfer copyright interests in the Documents, the copyright shall remain with Design-Builder.

3.1.9.3 USE OF DOCUMENTS IN EVENT OF TERMINATION In the event of a termination of this Agreement pursuant to ARTICLE 12, Owner shall have the right to use, to reproduce, and to make derivative works of the Documents to complete the Project, regardless of whether there has been a transfer of copyright under this section, provided payment has been made pursuant to §3.1.9.

3.1.9.4 OWNER'S USE OF DOCUMENTS AFTER COMPLETION OF PROJECT After completion of the Project, Owner may reuse, reproduce, or make derivative works from the Documents solely for the purposes of maintaining, renovating, remodeling, or expanding the Project at the Worksite. Owner's use of the Documents without Design-Builder's involvement or on other projects is at Owner's sole risk, except for Design-Builder's indemnification obligations, and Owner shall indemnify and hold harmless Design-Builder, its Design-Professional, Subcontractors, and consultants, and the agents, officers, directors, and employees of each of them, from and against any and all claims, damages, losses, costs, and expenses, including reasonable attorneys' fees and costs, arising out of or resulting exclusively from such any prohibited use.

3.1.9.5 DESIGN-BUILDER'S USE OF DOCUMENTS Where Design-Builder has transferred its copyright interest in the Documents under §3.1.9, Design-Builder may reuse Documents prepared by it pursuant to this Agreement in its practice, but only in their separate constituent parts and not as a whole.

3.1.9.6 Design-Builder shall obtain from its Design-Professional, Subcontractors, and consultants rights and rights of use that correspond to the rights given by Design-Builder to Owner in this Agreement, and Design-Builder shall provide evidence that such rights have been secured.

3.2 GUARANTEED MAXIMUM PRICE (GMP)

3.2.1 GMP PROPOSAL Owner and Design-Builder jointly agree, that Design-Builder shall perform the Work and complete the Work on a cost plus basis, for an amount not to exceed the Guaranteed Maximum Price (GMP) of one-hundred fourteen million, five-hundred thirty-three thousand eight hundred and fifty-three dollars (\$114,533,853.00) as set forth in **Exhibit 1**. Design-Builder represents that the GMP reflects the sum of the estimated Cost of the Work as defined in ARTICLE 8 as well as Design-Builder's Fee as defined in ARTICLE 7 along with Design-Builder's Contingency as provided in §3.2.44. The GMP is subject to modification as provided in ARTICLE 9, but is otherwise fixed and guaranteed by Design-Builder. However, Design-Builder does not guarantee any specific line item provided as part of the GMP, but agrees that it will be responsible for paying all costs of completing the Work which exceed the GMP, as adjusted in accordance with this Agreement.

3.2.1.1 To the extent the Current Drawings and Specifications are not complete at the time this Agreement is signed, Design-Builder shall provide for further development of the Design-Build Documents consistent with Owner's Program within the GMP. Such further development does not include (i) Owner directed changes in scope, systems, kinds and quality of materials, finishes, or equipment or (ii) items for which this Agreement authorizes or requires a change to the GMP, all of which, if required, shall be incorporated by Change Order.

3.2.2 BASIS OF GUARANTEED MAXIMUM PRICE

Design-Builder has based the GMP Proposal on this Agreement and the Contract Documents. Clarifications, Exclusions, and Contingencies are expressly stated therein.

3.2.2.1 Site Examination. Design-Builder hereby represents that it has conducted a thorough examination of the Worksite. Except for, and subject to, the Clarifications, Allowances, and Exclusions in the LNTP, and except as otherwise provided in this Agreement and Article 9.4, the failure of Design-Builder to fully familiarize itself with the foregoing conditions shall neither (a) provide any cost or schedule relief for Design-Builder or entitle Design-Builder to a Change under Article 9 nor (b) relieve Design-Builder of its obligations under this Agreement; however, for the avoidance of doubt, Design-Builder shall be reimbursed for any related expenses as, and to the extent, permitted by Article 8, and subject to the GMP. Notwithstanding the foregoing, Design-Builder is not responsible for offsite work performed by others that may affect utilities or access to the site or the progress of the Work ("Off Site Work"). The GMP and Project Schedule expressly exclude any costs or delays arising from Off Site Work. If Off Site Work impacts the progress or Cost of the Work, Design-Builder shall be entitled to an equitable adjustment in the GMP and/or the Contract Milestones.

3.2.2.2 Subsurface Conditions. Design-Builder has satisfied itself that the Worksite is satisfactory for performance of the Work. Except as provided in the LNTP, and except as provided in this Agreement and Article 9.4, the existence of any subsurface condition or any other condition on or under the Worksite shall neither (a) provide any cost or schedule relief for Design-Builder or entitle Design-Builder to a Change under Article 9 nor (b) relieve Design-Builder of its obligations under this Agreement; however, for the avoidance of doubt, (i) Design-Builder shall be reimbursed for any related expenses as, and to the extent, permitted by Article 8, and subject to the GMP and (ii) Design Builder shall be entitled to a Change under Article 9 to the extent an allowance in the LNTP is exceeded as further set forth in, and at the Unit Price included in, the LNTP.

3.2.2.3 Nature of the Work. Design-Builder has taken the necessary steps to understand the nature of the Work, the general local conditions that can affect the Work and the cost thereof including conditions relating to transportation, handling, storage of materials, Taxes, insurance, availability of qualified labor, unions, roads, weather, permits and licenses, applicable Laws and the character and availability of equipment and facilities needed preliminary to and during the

prosecution of the Work. Except as otherwise provided in the LNTP, this Agreement, and Article 6.3, these conditions shall neither (a) provide any cost or schedule relief for Design-Builder or entitle Design-Builder to a Change under Article 9 nor (b) relieve Design-Builder of its obligations under this Agreement; however, for the avoidance of doubt, Design-Builder shall be reimbursed for any related expenses as, and to the extent, permitted by Article 8, and subject to the GMP.

3.2.2.4 The Guaranteed Maximum Price includes all costs, expenses and fees of any kind or nature to be incurred by Design-Builder in connection with the performance of the Work (including all applicable taxes, duties, permitting costs, overhead (at any level), typical downtime of workers and machinery or equipment during the ordinary course of construction, fees and profit related thereto), and Design-Builder shall not be entitled to any additional compensation or reimbursement of any kind with respect to the performance of the Work, except in connection with any adjustment to the Guaranteed Maximum Price by a Change Order in accordance with the provisions of Article 9.

3.2.3 PRE-GMP WORK Any and all Work performed by Design-Builder shall be covered by the terms of this Agreement, and included within the Guaranteed Maximum Price, regardless of whether such Work was performed prior to or after the execution of this Agreement or originally pursuant to the LNTP.

3.2.4 DESIGN-BUILDER'S CONTINGENCY The GMP Proposal contains, as part of the estimated Cost of the Work, the Design-Builder's Contingency (Design/Bid Contingency and CM Risk/Construction Contingency), which is a sum mutually agreed upon and monitored by Design-Builder and Owner to cover costs which are properly reimbursable as a Cost of the Work but are not the basis for a Change Order. Design-Builder's Contingency shall not be used for changes in scope or for any item that would be the basis for an increase in the GMP. Design-Builder shall provide Owner with a monthly accounting of charges against Design-Builder's Contingency, if applicable, with each application for payment.

3.2.5 COST REPORTING Design-Builder shall maintain complete and current records that comply with generally accepted accounting principles and calculate the Cost of Work. Owner shall be afforded access during normal business hours to all Design-Builder's records, books, correspondence, instructions, drawings, receipts, vouchers, memoranda, and similar data relating to this Agreement. Design-Builder shall preserve all such records for a period of three years after the final payment or longer where required by Law.

3.3 CONSTRUCTION PHASE SERVICES

3.3.1 Design-Builder's obligations for the Construction Phase include:

3.3.1.1 Clearance and Preparation of Site; Temporary Facilities <u>Clearance and Preparation of Site</u>. Design-Builder shall be responsible for clearing the Worksite, including the removal of both natural and man-made obstructions and performing all things necessary to begin construction. Design-Builder shall provide for the procurement of, as appropriate, all soil, gravel and similar materials required for performance of the Work.

3.3.1.1.2 <u>Temporary Facilities</u>. Design-Builder shall be responsible for providing all temporary facilities and utilities necessary for the performance of the Work under this Agreement, including water, sewage, electric power, telephone service, access roads, laydown areas and internet connections. Such facilities shall be subject to the review and comment of Owner and shall include, but not be limited to, temporary perimeter fencing, temporary controlled access points, first aid and hazardous

response personnel and facilities, security personnel, climate-controlled offices (including for use by Owner's representatives, and Owner's Nominees), workshops, restrooms and warehouses and all required temporary utilities. Design-Builder shall arrange and pay for disposal of sewage and wastes as necessary to enable Design-Builder to perform the Work. Design-Builder shall provide adequate protection against, and treatment of, water runoff from the Worksite in accordance with all aapplicable Laws.

3.3.1.1.3 <u>Owner's Working Accommodations</u>. Design-Builder shall, upon request by Owner from time to time, provide Owner, Owner's Nominees, and its designees with furnished office space with all of the infrastructure described in Section 3.3.1.1.2 for up to ten (10) people at the Worksite.

3.3.1.2 Procurement and Storage of Materials, Equipment and Supplies Generally. Design-Builder shall procure, pay for, receive, and transport to the Worksite, store, warehouse and maintain according to manufacturers' recommendations (or, if such recommendations do not exist, in accordance with Good Industry Practices) and taking into account foreseeable environmental conditions, as an independent contractor and not as agent for Owner, all materials, equipment (including temporary equipment), supplies, machinery, tools, and all labor and manufacturing and related services necessary or desirable for construction, start-up, commissioning of the Work, whether or not incorporated into the Project, that are required for completion of the Work in accordance with this Agreement except those items specified as being furnished by Owner pursuant to Article 4 and the Contract Documents. Design-Builder shall, promptly upon receipt from any Subcontractor, furnish copies of such Subcontractor's equipment manuals and storage specifications to Owner.

<u>Chemicals and Consumables</u>. Design-Builder shall arrange and pay for all consumables required for Design-Builder to perform the Work. Design-Builder shall also arrange and pay the disposal of all such unused consumables no longer needed for the Work. Design-Builder shall keep a complete inventory, with material safety data sheets, of all chemical and other related materials (including Hazardous Substances) brought onto any Worksite for any purpose.

- 3.3.1.2.3 <u>Construction Power</u>. Design-Builder, at its sole cost and expense, shall be responsible for securing temporary electrical power for the performance of the Work. Owner shall be responsible for securing rights-of-way or easements necessary for temporary power.
- 3.3.1.2.4 <u>Construction Water & Temporary Utilities</u>. Design-Builder shall be solely responsible for providing and paying for any and all quantities of water, including potable water, temporary industrial water, fire protection water and wastewater disposal, or other temporary utilities including the installation and removal of any temporary distribution systems, required by Design-Builder for or in connection with the performance of the Work.

3.3.1.3 Labor and Personnel. <u>Generally</u>. Design-Builder shall provide and be solely responsible for all labor and personnel required in connection with the Work, including: (a) professional engineers, as applicable, licensed to perform engineering services, or their appropriate designee, at the Worksite and qualified to perform the type of engineering services required of Design-Builder hereunder; and (b) a project manager who shall have the experience and authority to administer this Agreement on behalf of Design-Builder to furnish information as reasonably requested by Owner and to agree upon procedures for coordinating Design-Builder's efforts with those of Owner, and who shall be present, unless otherwise agreed by Owner, at the Worksite at all times when the Work is being performed at the Worksite.

- 3.3.1.3.2 <u>Labor</u>. Design-Builder shall also be responsible for any and all immigration or other labor, permits, licenses, taxes, and other requirements or limitations under applicable Law governing its Work.
- 3.3.1.3.3 <u>Key Personnel</u>. Design-Builder's staff shall include the key personnel set forth in <u>Exhibit 2</u>, who shall be solely dedicated to the Project and shall be acceptable to Owner (the "Key Personnel"). Except for reasons requiring immediate termination, Design-Builder shall not remove or replace any of the Key Personnel without (i) the prior written consent of Owner and (ii) providing for an appropriate transfer of responsibilities to another qualified person approved by Owner. If at any time during the performance of the Work any of Design-Builder's Key Personnel becomes unacceptable to Owner for any reason, then, upon notice from Owner, Design-Builder shall promptly replace such individual with an individual acceptable to Owner. If at any time during the performance of the Work any of the Key Personnel should no longer be available to perform services in connection with the Work, then Design-Builder shall promptly replace such individual with an individual with an individual acceptable to Owner whose qualifications and experience are equal to or better than those of the replaced individual.

3.3.1.4 Permitting.

- 3.3.1.4.1 <u>Generally</u> Design-Builder shall obtain, pay for and maintain in effect all construction permits, licenses, approvals and other government authorizations ("Permits") required under applicable Laws in connection with design, engineering and construction of the Project or performance of the Work, excluding any zoning or land use approvals which shall remain Owner's responsibility. Owner shall cooperate and provide authorization, after review, for Design-Builder to obtain any Permits which are required to be issued in Owner's name. Owner is responsible for obtaining any zoning approval. Design Builder is not currently aware of any zoning or land use approvals which Owner needs to obtain for Design Builder to perform the Work, except for potential issues with the outdoor shooting range, which Owner is handling.
- 3.3.1.4.2 <u>Deliverables</u>. No later than sixty (60) Days after the execution of this Agreement, Design-Builder shall submit to Owner: (A) evidence demonstrating that Design-Builder has obtained all Permits then required to be obtained for the performance of the Work; (B) a list identifying the Permits not yet required to be obtained for the performance of the Work; and (C) a list identifying the Permits applied for by Design-Builder but not yet issued or received, together with a plan reasonably acceptable to the Owner for obtaining such Permits and an estimate of the time within which such Contractor Permits will be obtained.

3.3.1.4.3 <u>Compliance and Closure</u>. Design-Builder shall keep Owner informed of the progress it is making in obtaining and maintaining all Permits and provide Owner with copies of any Permits and any correspondence pertaining thereto within one (1) Business Day of obtaining such Permit(s). Design-Builder hereby agrees, acknowledges and understands that it is solely responsible for obtaining, maintaining and complying with all of the Permits. As soon as available, Design-Builder shall provide to Owner a copies of all Permits that have been issued to Design-Builder and/or received by Design-Builder on behalf of Owner for the Work, and documentation that they have been satisfied or otherwise closed by the applicable governmental authority.

3.3.1.5 Protection of Property, Materials and Equipment. No person performing Work for or on behalf of Design-Builder shall damage any public or private property (including all parallel, converging and intersecting electric lines and poles, telephone or cable lines and poles, roads, highways, waterways, railroads, sewer lines, natural gas pipelines, drainage ditches, culverts, buildings, structures, monuments, vehicles, trees, shrubs, lawns, walks and pavements) on roads and routes to the Worksite or along, adjacent to, or near the Worksite. If any person performing Work for or on behalf of Design-Builder directly or indirectly causes any damage to any such property, then Design-Builder shall immediately notify Owner and, at Design-Builder's own expense, repair, rebuild, reinforce or otherwise fully restore the same to the satisfaction of the property owner and in accordance with all applicable Laws.

3.3.2 In order to complete the Work, Design-Builder shall provide all necessary construction supervision, inspection, construction equipment, labor, materials, tools, and subcontracted items.

3.3.3 COMPLIANCE WITH LAWS Design-Builder shall comply with all Laws at its own costs. Design-Builder shall be liable to Owner for all loss, cost, and expense, attributable to any acts or omissions by Design-Builder, its employees, subcontractors, and agents resulting from the failure to comply with Laws, including fines, penalties, or corrective measures. However, liability under this subsection shall not apply if notice to Owner was given, and advance approval by appropriate authorities, including Owner, is received.

3.3.3.1 CHANGES IN LAW The GMP, Design-Builder's Fee, and Contract Milestones shall be equitably adjusted by Change Order pursuant to ARTICLE 9 for additional costs and time resulting from any changes in Laws, including increased taxes, which were enacted after the date of this Agreement, or in the case of the GMP, after the date of an accepted GMP proposal.

3.3.4 Monthly Progress Meetings. On a monthly basis or more frequently if Owner determines that Project conditions require it, Design-Builder shall hold progress meetings with Owner, Owner's authorized representatives and their respective invitees on a monthly basis (each, a "Monthly Progress Meeting") or more frequently (including weekly or daily) if required by the status of the Work or as reasonably requested by Owner. Design-Builder's General Manager and Project Manager as well as the lead architect shall attend the Monthly Progress Meeting.

3.3.5 Design-Builder shall provide monthly written reports to Owner on the progress of the Work in such detail as is reasonably agreed to by the Parties (each, a "Monthly Progress Report"). At minimum, each Monthly Progress Report shall describe the actual progress of the Work, include a compliant Current Project Schedule, and Recovery Plan, if applicable, and updated Cost Report. Each Monthly Progress Report shall be submitted to Owner on a monthly basis and or before the Monthly Progress Meeting. The delivery of each Monthly Progress Report shall be a condition precedent to Owner's obligation to make the corresponding Progress Payment.

3.3.6 Design-Builder shall develop a system of cost reporting for the Work, including monthly monitoring of actual costs for activities in progress and estimates for uncompleted tasks and proposed changes in the Work (each, a "Cost Report"). A Cost Report shall be presented to Owner on a monthly basis and or before the Monthly Progress Meeting. The delivery of each Cost Report shall be a condition precedent to Owner's obligation to make the corresponding Progress Payment.

3.3.7 Design-Builder shall regularly remove debris and waste materials at the Worksite resulting from the Work. Prior to discontinuing Work in an area, Design-Builder shall clean the area and remove all rubbish and its construction equipment, tools, machinery, waste, and surplus materials. Design-Builder shall minimize and confine dust and debris resulting from construction activities. At the completion of the Work, Design-Builder shall remove from the Worksite all construction equipment, tools, surplus materials, waste materials, and debris.

3.3.8 Design-Builder shall prepare and submit to Owner either:

□ final marked-up as-built drawings, or

I updated electronic data along with a redline .pdf of the as-built drawings

 \Box such documentation as defined by the Parties by attachment to this Agreement, in general documenting how the various elements of the Work were actually constructed or installed.

3.3.9 Owner's Instructions. Design-Builder shall immediately comply with and strictly adhere to all of Owner's instructions and directions given by Owner's Authorized Representative (whether or not Design-Builder disputes or questions such instructions or directions), except to the extent (a) prevented by any applicable Law or (b) Design-Builder reasonably believes that compliance with such instructions or directions would cause safety hazards to any Person or the environment. If Design-Builder believes that any Owner instruction constitutes a Change, Design-Builder shall comply with Article 9. If Design-Builder disputes any instruction or direction of Owner, such dispute shall be resolved in accordance with Article 13; provided that Design-Builder shall comply with all instructions and directions received from Owner and shall continue to perform the Work pending resolution of the dispute. Design-Builder shall take instructions and directions only from Owner in writing as to the performance of the Work.

3.3.10 Standards of Performance for Construction Phase Work. Design-Builder shall perform the work:

- (a) in a safe, timely, professional, prudent, economical, expeditious and workmanlike manner and with the degree of skill and care that is utilized by professionals in the same field under the same or similar circumstances in accordance with Good Industry Practices;
- (b) in accordance and compliance with (i) this Agreement, (ii) the requirements contained in the Owner's Program, (iii) all applicable Laws, (iv) Good Industry Practices, including applicable safety, environmental, construction and electrical codes, standards, as such codes and standards exist (v) Design-Builder's Construction Documents, and (vi) the Permits;
- (c) in complies with accordance with all applicable Laws that are in effect on the date of Substantial Completion of the Work;
- (d) using only new materials and workmanship that conform to the Construction Documents;
- (e) such that the Work will conform to the Contract Documents;
- (f) such that the Work (excluding Services) is free from defects, whether latent or patent, in workmanship or materials;
- (g) relating to Services (i) in accordance with this Agreement and (ii) consistent with the Design Standard of Care; and
- (h) in a manner that efficiently utilizes Owner's materials and resources

(collectively, the "Standards of Performance").

3.4 SCHEDULING

3.4.1 CONSTRUCTION SCHEDULE Design-Builder shall prepare and submit the Construction Schedule setting forth Design-Builder's plan for completing the Work for Owner's written approval consistent with Good Industry Practices. This Construction Schedule shall comply with the following:

3.4.1.1 <u>General Scheduling Requirements</u>. The Construction Schedule, and as-updated the Current Construction Schedule, shall provide detailed monitoring of the progress of the Work and provide the Parties with a tool to monitor and follow the progress of all phases of the Work, the scheduling and progress monitoring of all Subcontractors, the scheduling for any applicable third party activities and reporting for the Owner and other stakeholders. Design-Builder shall plan in detail each area of the Work following the requirements of this Agreement. All portions of the Work shall be scheduled in sufficient detail to enable a correct logical sequence of activities to be established. Approvals and review periods of Owner and other persons, if any, shall be included. The Construction Schedule for the Work shall:(a) be consistent with the Project Schedule; (b) include the requirements of this Agreement; (c) remain unchanged throughout the execution of the Work unless adjusted pursuant to a Change Order; (d) be the schedule against which all Work progress is measured; and (e) begin with the issuance of the NTP and include Key Milestones.

3.4.1.2 <u>Schedule Detail</u>. All engineering, Worksite and Worksite-related activity durations, including transportation and transportationrelated activities, for all schedules shall be broken into sufficient detail and shall generally not exceed two (2) weeks; activities exceeding this duration shall be broken into sub-activities. Activities shall include all aspects of the Work for which Design-Builder and all of its Subcontractors are responsible. Activities shall be broken down sufficiently for the information provided to service the overall Project Schedule and Current Construction Schedule.

3.4.1.3 <u>Key Milestones</u>. Contract Milestones shall be included in the Project Schedule and Current Project Schedule, as well as the inspection and witness points and testing dates (collectively, the "Key Milestones"). The Parties shall agree on the identification and timing of other milestones that shall be included in the Project Schedule and Current Construction Schedule.

3.4.1.4 <u>Owner's Review of Logic & Dates</u>. Design-Builder has represented that it will achieve each of the Contract Milestones by the dates set forth in the Project Schedule and Owner has accepted those dates which shall be incorporated into the Construction Schedule. Owner however makes no representation regarding the reasonableness or sufficiency of the durations or inherent sequencing included in the Project or Construction Schedule and Design-Builder alone shall be responsible for any failure to comply with or achieve the Contract Milestones.

3.4.2 The Current Construction Schedule. The Construction Schedule shall be updated with the current status of the Work on a monthly basis, without the addition of any logic changes or activities, unless approved by Change Order (the "Current Construction Schedule"). All Work progress shall be measured by comparing actual progress as indicated in the Current Project Schedule to progress required by the Project Schedule. The delivery of the Current Construction Schedule each month shall be a condition precedent to Owner's obligation to make the corresponding Progress Payment.

3.4.3 Recovery Plan.

3.4.3.1 If, at any time during the performance of the Work (whether as indicated in Design-Builder's Monthly Progress Report, the Current Construction Schedule, or as otherwise communicated to Owner or as otherwise becomes known to Design-Builder or Owner at any time during the performance of the Work):

3.4.3.1.1 Design-Builder fails, or believes that it will fail, to achieve any of the Contract Milestones; or



- 3.4.3.1.2 Design-Builder is, or believes that it will be, more than thirty (30) Days behind any of the Contract Milestone Dates, at any time during the performance of the Work up to thirty (30) Days prior to the Guaranteed Completion Date for Substantial Completion of the Work; or
- 3.4.3.1.3 Design-Builder is, or believes that it will be more than seven (7) Days behind any of the dates in the Project Schedule (other than any Contract Milestone) at any time during the last thirty (30) Days prior to the Guaranteed Completion Date for Substantial Completion of the Work,

Design-Builder shall, in accordance with this Section, immediately prepare a Recovery Plan to regain compliance with the Construction Schedule. Specifically, Design-Builder shall, (a) participate in such meetings together those Subcontractors as Owner may reasonably request, to discuss and illustrate how Design-Builder intends to regain compliance with the Project Schedule and (b) develop and implement a Recovery Plan demonstrating how it will regain compliance with the Project Schedule to Owner as soon as practicable, but in all cases within seven (7) Days after receiving notification from Owner. Upon receipt of a Recovery Plan, Owner will review and may comment upon the same. Design-Builder shall accept and incorporate all Owner's comments and resubmit to Owner, within seven (7) Days after receiving such comments, the revised Recovery Plan to Owner. Design-Builder shall in all cases immediately implement the Recovery Plan (or such parts thereof as are possible at that stage) whether or not the review process described above has been completed. Where Design-Builder is the cause, or to the extent the concurrent cause, of the delay in the Project Schedule that necessitates the Recovery Plan, Design-Builder shall implement the Recovery Plan at its own cost and without any adjustment of the Guaranteed Completion Dates.

3.4.3.2 Whenever Design-Builder is required to implement a Recovery Plan in order to regain compliance with the Construction Schedule, it shall do so by utilizing additional shifts, additional manpower, additional or different construction equipment, more efficient work methods, overtime and/or re-sequencing of activities. During the course of implementing any Recovery Plan, Design-Builder, at Owner's request, shall conduct scheduling meetings, as frequently as Owner may require, with Owner and any Subcontractors that Owner may require, for the purpose of discussing status of the planning and effectiveness of the Recovery Plan.

3.4.3.3 Nothing in this Section shall relieve Design-Builder of any of its obligations under this Agreement. Where Design-Builder is the cause, or to the extent the concurrent cause, of the delay in the Project Schedule that necessitates the Recovery Plan, Design-Builder shall bear its own costs and expenses in connection with the preparation and implementation of a Recovery Plan.

3.4.4 Design-Builder shall perform the Work in accordance with the Project Schedule, the Construction Schedule, the Current Construction Schedule and, where necessary, any Recovery Plan.

3.5 SAFETY OF PERSONS AND PROPERTY

3.5.1 SAFETY PRECAUTIONS AND PROGRAMS Design-Builder shall have overall responsibility for safety precautions and programs in the performance of the Work. However, such obligation does not relieve Subcontractors of their responsibility for the safety of persons or property in the performance of their work, nor for compliance with the provisions of Laws.

3.5.2 Design-Builder shall prevent against injury, loss, or damage to persons or property by taking reasonable steps to protect:

3.5.2.1 its employees and other persons at the Worksite;

3.5.2.2 materials, supplies, and equipment stored at the Worksite for use in performance of the Work; and

3.5.2.3 the Project and all property located at the Worksite and adjacent to work areas, whether or not said property or structures are part of the Project or involved in the Work.

3.5.3 DESIGN-BUILDER'S SAFETY REPRESENTATIVE Design-Builder shall designate an individual at the Worksite in the employ of Design-Builder who shall act as Design-Builder's designated safety representative with a duty to prevent accidents. Unless otherwise identified by Design-Builder in writing to Owner, the designated safety representative shall be Design-Builder's project superintendent. Design-Builder will report promptly in writing all recordable accidents and injuries occurring at the Worksite to Owner. When Design-Builder is required to file an accident report with a public authority, Design-Builder shall furnish a copy of the report to Owner.

3.5.4 Design-Builder shall provide Owner with copies of all notices required of Design-Builder by Law. Design-Builder's safety program shall comply with the requirements of governmental and quasi-governmental authorities having jurisdiction over the Work.

3.5.5 Damage or loss not insured under property insurance that arises from the performance of the Work, to the extent of the negligence attributed to such acts or omissions of Design-Builder, or anyone for whose acts Design-Builder may be liable, shall be promptly remedied by Design-Builder. Damage or loss attributable to the acts or omissions of Owner or Others and not to Design-Builder shall be promptly remedied by Owner.

3.5.6 If Owner deems any part of the Work or Worksite unsafe, Owner, without assuming responsibility for Design-Builder's safety program, may require Design-Builder to stop performance of the Work or take corrective measures satisfactory to Owner, or both. If Design-Builder does not adopt corrective measures, Owner may perform them and reduce by the costs of the corrective measures the amount of the GMP, or in the absence of a GMP, the Cost of the Work as provided in ARTICLE 8. Design-Builder agrees to make no claim for damages, for an increase in the GMP, compensation for Design Phase services, Design-Builder's Fee, or the Contract Milestones based on Design-Builder's compliance with Owner's reasonable request.

3.6 EMERGENCIES In any emergency affecting the safety of persons or property, Design-Builder shall act in a reasonable manner to prevent threatened damage, injury, or loss. Any change in the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contract Milestones, and if appropriate the compensation for Design Phase services, on account of emergency work shall be determined as provided for in ARTICLE 9.

3.7 HAZARDOUS MATERIALS

3.7.1 Design-Builder shall not be obligated to commence or continue a portion of the Work impacted by Hazardous Materials at the Worksite until such Hazardous Material has been removed, rendered, or determined to be harmless by Owner as certified by an independent testing laboratory and approved by the appropriate governmental agency.

3.7.2 If, after commencing the Work, Hazardous Material is discovered at the Project, Design-Builder shall be entitled to immediately stop Work in the affected area. Design-Builder shall promptly report the condition to Owner and, if required, the governmental agency with jurisdiction.

3.7.3 Design-Builder shall not resume nor be required to continue any Work affected by any Hazardous Material without written mutual agreement between the Parties after the Hazardous Material has been removed or rendered harmless and only after approval, if necessary, of the governmental agency with jurisdiction.

3.7.4 Owner shall be responsible for retaining an independent testing laboratory to determine the nature of the material encountered and whether it is a Hazardous Material requiring corrective measures or remedial action. Such measures shall be the sole responsibility of Owner, and shall be performed in a manner minimizing any adverse effect upon the Work.

3.7.5 If Design-Builder incurs additional costs or is delayed due to the presence or remediation of Hazardous Material discovered at the Worksite, Design-Builder shall be entitled to an equitable adjustment in the GMP, compensation for Design Phase services, Design-Builder's Fee, or the Date of Substantial Completion or the Date of Final Completion.

3.7.6 To the extent permitted under §6.5 and to the extent not caused by the negligent or intentionally wrongful acts or omissions of Design-Builder, its Subcontractors, Suppliers, and Sub-subcontractors, and the agents, officers, directors, and employees of each of them, Owner shall defend, indemnify, and hold harmless Design-Builder, its Subcontractors, and Subsubcontractors, and the agents, officers, directors, and employees of each of them, from and against all claims, damages, losses, costs, and expenses, including but not limited to reasonable attorneys' fees, costs, and expenses incurred in connection with any dispute resolution procedure, arising out of or relating to the performance of the Work in any area affected by Hazardous Material.

3.7.7 MATERIALS BROUGHT TO THE SITE

3.7.7.1 Safety Data Sheets (SDS) as required by Law and pertaining to materials or substances used or consumed in the performance of the Work, whether obtained by Design-Builder, Subcontractors, Owner, or Others, shall be maintained at the Project by Design-Builder and made available to Owner and Subcontractors.

3.7.7.2 Design-Builder shall be responsible for the proper handling, application, storage, removal, and disposal of all materials brought to the Worksite by Design-Builder. Upon the issuance of the Certificate of Substantial Completion of the Work, Owner shall be responsible for materials and substances brought to the Worksite by Design-Builder if such materials or substances are required by the Contract Documents.

3.7.7.3 To the extent caused by the negligent or intentionally wrongful acts or omissions of Design-Builder, its agents, officers, directors, and employees, Design-Builder shall indemnify and hold harmless Owner, its agents, officers, directors, and employees, from and against any and all claims, damages, losses, costs, and expenses, including but not limited to attorneys' fees, costs, and expenses incurred in connection with any dispute resolution procedure, arising out of or relating to the delivery, handling, application, storage, removal, and disposal of materials and substances brought to the Worksite by Design-Builder.

3.7.8 Section 3.7 shall survive the completion of the Work or Agreement termination.

3.8 WARRANTY

3.8.1 Design-Builder warrants that all materials and equipment furnished under the Construction Phase of this Agreement will be new unless otherwise specified, of good quality, in conformance with the Contract Documents, and free from defective workmanship and materials. Warranties shall commence on the Date of Substantial Completion of the Work or of a designated portion.

3.8.2 To the extent products, equipment, systems, or materials incorporated in the Work are specified and purchased by Owner, they shall be covered exclusively by the warranty of the manufacturer. There are no warranties which extend beyond the description on the face of any such warranty. To the extent products, equipment, systems, or materials incorporated in the Work are specified by Owner but purchased by Design-Builder and are inconsistent with selection criteria that otherwise would have been followed by Design-Builder, Design-Builder shall assist Owner in pursuing warranty claims. ALL OTHER WARRANTIES EXPRESSED OR IMPLIED INCLUDING THE WARRANTY OF MERCHANTABILITY AND THE WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE ARE EXPRESSLY DISCLAIMED.

3.8.3 Design-Builder shall secure required certificates of inspection, testing, or approval and deliver them to Owner.

3.8.4 Design-Builder shall collect all written warranties and equipment manuals and deliver them to Owner in a format directed by Owner.

3.8.5 With the assistance of Owner's maintenance personnel, Design-Builder shall direct the checkout of utilities and start-up operations, and adjusting and balancing of systems and equipment for readiness.

3.9 CORRECTION OF WORK WITHIN ONE YEAR

3.9.1 Before Substantial Completion and within one year after the date of Substantial Completion of the Work or for such longer periods of time as may be set forth with respect to specific warranties required by the Contract Documents, if any Defective Work is found, Owner shall promptly notify Design-Builder in writing. Unless Owner provides written acceptance of the condition, Design-Builder shall promptly correct the Defective Work at its own cost and time and bear the expense of Additional Services required for correction of any Defective Work for which it is responsible. If within the one-year correction period Owner discovers and does not promptly notify Design-Builder or give Design-Builder an opportunity to test or correct Defective Work as reasonably requested by Design-Builder, Owner waives Design-Builder's obligation to correct that Defective Work as well as Owner's right to claim a breach of the warranty with respect to that Defective Work.

3.9.2 With respect to any portion of Work first performed after Substantial Completion, the one-year correction period shall commence when that portion of Work is complete. Correction periods shall not be extended by corrective work performed by Design-Builder.

3.9.3 If Design-Builder fails to correct Defective Work within a reasonable time after receipt of written notice from Owner before final payment, Owner may correct it in accordance with Owner's right to carry out the Work. In such case, an appropriate Change Order shall be issued deducting the cost of correcting the Defective Work from payments then or thereafter due Design-Builder. If payments then or thereafter due Design-Builder are not sufficient to cover such amounts, Design-Builder shall pay the difference to Owner.

3.9.4 Design-Builder's obligations and liability, if any, with respect to any Defective Work discovered after the one-year correction period shall be determined by the Law. If, after the one-year correction period but before the applicable limitation period has expired, Owner discovers any Work which Owner considers Defective Work, Owner shall, unless the Defective Work requires emergency correction, promptly notify Design-Builder and allow Design-Builder an opportunity to correct the Work if Design-Builder elects to do so. If Design-Builder elects to correct the Work, it shall provide written notice of such intent within fourteen (14) Days of its receipt of notice from Owner and shall complete the correction of Work within a mutually agreed timeframe. If Design-Builder does not elect to correct the Work, Owner may have the Work corrected by itself or others, and, if Owner intends to seek recovery of those costs from Design-Builder, Owner shall promptly provide Design-Builder with an accounting of actual correction costs.

3.9.5 If Design-Builder's correction or removal of Defective Work causes damage to or destroys other completed or partially completed Work or existing buildings, Design-Builder shall be responsible for the cost of correcting the destroyed or damaged property.

3.9.6 The one-year period for correction of Defective Work does not constitute a limitation period with respect to the enforcement of Design-Builder's other obligations under the Contract Documents.

3.9.7 Before final payment, at Owner's option and with Design-Builder's agreement, Owner may elect to accept Defective Work rather than require its removal and correction. In such case the Guaranteed Maximum Price shall be equitably adjusted for any diminution in the value of the Project caused by such Defective Work.

3.10 CONFIDENTIALITY Design-Builder shall treat as confidential and not disclose to third-persons, nor use for its own benefit ("Treat as Confidential"), any of Owner's confidential information, know-how, discoveries, production methods, and the like disclosed to Design-Builder or which Design-Builder may acquire in performing the Work. To the extent necessary to perform the Work, Design-Builder's confidentiality obligations do not apply to disclosures to Subcontractors, Subsubcontractors, and Suppliers. Owner shall Treat as Confidential information all of Design-Builder's estimating systems and historical and parameter cost data disclosed to Owner in performing the Work. Except for information that Owner obtains through ownership of the copyright, Owner shall treat as confidential information all design systems that may be disclosed to Owner in connection with the performance of this Agreement. Each Party shall specify and mark confidential items as "Confidential." Confidentiality obligations do not supersede compulsion by Law, a governmental agency or authority, an order of a court of competent jurisdiction, or a validly issued subpoena. In such event, a Party shall promptly notify the other Party to permit that Party's legal objection.

3.11 Omitted.

3.12 DESIGN-BUILDER'S REPRESENTATIVE Design-Builder shall designate a person who shall be Design-Builder's representative. Design-Builder's Representative is Marty E. Gibbs, Senior Vice President & General Manager ([marty.gibbs@christmanco.com & 865-389-3109])

ARTICLE 4 OWNER'S RESPONSIBILITIES

4.1 INFORMATION AND SERVICES PROVIDED BY OWNER Owner's responsibilities under this article shall be fulfilled with reasonable detail and in a timely manner.

4.2 FINANCIAL INFORMATION Before commencing the Work and thereafter, at the written request of Design-Builder, Owner shall provide Design-Builder evidence of Project financing. Evidence of such financing shall be a condition precedent to Design-Builder's commencing or continuing the Work. Design-Builder shall be notified before any material change in Project financing.

4.3 WORKSITE INFORMATION To the extent Owner has possession of the following Owner Provided Information it has provided it to Design-Builder.

4.3.1 information describing the physical characteristics of the site, including surveys, Worksite evaluations, legal descriptions, data, or drawings depicting existing conditions, subsurface conditions, and environmental studies, reports, and investigations;

4.3.2 tests, inspections, and other reports dealing with environmental matters, Hazardous Material, and other existing conditions, including structural, mechanical, and chemical tests, required by the Contract Documents or by Law;

4.3.3 any other information or services requested in writing by Design-Builder which are required for Design-Builder's performance of the Work and under Owner's exclusive control.

The Owner Provided Information includes the September 29, 2021 C2RL Inc. Report.

4.4 MECHANICS AND CONSTRUCTION LIEN INFORMATION Within seven (7) Days after receiving Design-Builder's written request, Owner shall provide Design-Builder with the information necessary to give notice of or enforce mechanics lien rights and, where applicable, stop notices. This information shall include Owner's interest in the real property interests of the Worksite and the record legal title.

4.5 RESPONSIBILITIES DURING DESIGN PHASE

4.5.1 Owner shall review and timely provide written approvals required by this Agreement.

4.6 RESPONSIBILITIES DURING CONSTRUCTION PHASE

4.6.1 Owner shall review request for a Change Order, based on a Change and timely approve the same, if or as warranted.

4.6.2 If Owner becomes aware of any error, omission, or failure to meet the requirements of the Contract Documents or any fault or defect in the Work, Owner shall give prompt written notice to Design-Builder. The failure of Owner to give such notice shall not relieve Design-Builder of its obligations to fulfill the requirements of the Contract Documents.

4.6.3 Owner shall have no contractual obligations to Subcontractors, suppliers, or the Design-Professional.

4.6.4 Owner shall provide insurance for the Project as provided in ARTICLE 11.

4.7 TAX EXEMPTION If in accordance with Owner's direction Design-Builder claims an exemption for taxes, Owner shall indemnify and hold Design-Builder harmless for all liability, penalty, interest, fine, tax assessment, attorneys' fees, or other expense or cost incurred by Design-Builder as a result of any action taken by Design-Builder in accordance with Owner's direction.

4.8 ELECTRONIC DOCUMENTS If Owner requires that the Parties exchange documents and data in electronic or digital form, before any such exchange, Owner and Design-Builder shall agree on and follow ConsensusDocs 200.2 or a separate written protocol addendum governing all exchanges, which specifies: (a) the definition of documents and data to be accepted in electronic or digital form or to be transmitted electronically or digitally; (b) management and coordination responsibilities; (c) necessary equipment, software and services; (d) acceptable formats, transmission methods, and verification procedures; (e) methods for maintaining version control; (f) privacy and security requirements; and (g) storage and retrieval requirements. The Parties shall each bear their own costs for the requirements identified in the protocol. In the absence of a written protocol, use of documents and data in electronic or digital form shall be at the sole risk of the recipient.

4.9 OWNER'S REPRESENTATIVE Owner's Representative is Luc Hardyn, Vice President, Operations (Email: LHardyn@smith-wesson.com and Phone: (413) 747-6253)—. Owner's representative shall: (a) be fully acquainted with the Project; (b) agree to furnish the information and services required of Owner in a timely manner; and (c) have the authority to bind Owner in all matters requiring Owner's approval, authorization, or written notice of any immaterial change; however written authorization from Owner will be required for any material change. If Owner changes its representative or the representative's authority as listed above, Owner shall notify Design-Builder in writing in advance.

ARTICLE 5 SUBCONTRACTS

5.1 RETAINING SUBCONTRACTORS Design-Builder shall not retain any Subcontractor to whom Owner has a reasonable and timely objection, provided that Owner agrees to compensate Design-Builder for any additional costs incurred by Design-Builder as a result of such objection. Owner may propose subcontractors to be considered by Design-Builder. Design-Builder shall not be required to retain any subcontractor to whom Design-Builder has a reasonable objection.

5.2 MANAGEMENT OF SUBCONTRACTORS Design-Builder shall be responsible for the management of Subcontractors in the performance of their work.



5.3 CONTINGENT ASSIGNMENT OF SUBCONTRACT

5.3.1 If this Agreement is terminated, each subcontract agreement shall be assigned by Design- Builder to Owner, subject to the prior rights of any surety, provided that:

5.3.1.1 this Agreement is terminated by Owner pursuant to §0 or §12.4; and

5.3.1.2 Owner accepts such assignment, after termination by notifying in writing Design-Builder and Subcontractor or Design-Builder and Supplier, and assumes all rights and obligations of Design-Builder pursuant to each subcontract or supply agreement.

5.3.2 If Owner accepts such an assignment, and the Work has been suspended for more than thirty (30) consecutive Days, following termination, if appropriate, Subcontractor's or Supplier's compensation shall be equitably adjusted as a result of the suspension.

5.4 BINDING OF SUBCONTRACTORS AND SUPPLIERS Design-Builder agrees to bind every Subcontractor and Material Supplier (and require every Subcontractor to so bind its subcontractors and significant suppliers) to all the provisions of this Agreement and the Contract Document's applicable provisions to that portion of the Work.

ARTICLE 6 TIME

6.1 DATE OF COMMENCEMENT The Date of Commencement is the Effective Date in ARTICLE 1. The Work shall proceed in general accordance with the approved Project Schedule which may be amended from time to time, subject, however, to other provisions of this Agreement.

6.2 SUBSTANTIAL/FINAL COMPLETION, subject to adjustments as provided for in the Contract Documents, the Date of Substantial Completion, the Date of Final Completion, and the other Contract Milestones listed below (collectively "Contract Milestones") shall be completed by the following Guaranteed Completion Dates:

Contract Milestone	Guaranteed Completion Date	(aily Rate in USD\$) of lated Damages
Equipment Ready/Early Turnover – Area 4 (Assembly).			
Equipment Ready/Early Turnover means when the Owner can			
deliver and set equipment and excludes energization of			
equipment.	January 25, 2023	\$	1,500.00
See <u>Exhibit 1</u>			
Substantial Completion, excluding commissioning – Plant &			
Indoor Gun Range	June 13, 2023	\$	1,500.00
See <u>Exhibit 1</u>			
Substantial Completion of the Work, excluding commissioning,			
landscaping	September 1, 2023	\$	1,500.00

6.2.1 The deadlines for the Contract Milestones are subject to adjustments as provided for in the Contract Documents.

6.2.2 Time is of the essence with regard to the obligations of the Contract Documents.

6.2.3 Unless instructed by Owner in writing, Design-Builder shall not knowingly commence the Work before the effective date of insurance required to be provided by Design-Builder.

6.2.4 Liquidated damages shall not be compounded across Milestone Dates and shall be assessed non-cumulatively.

6.3 DELAYS AND EXTENSIONS OF TIME

6.3.1 If Design-Builder is delayed at any time in the commencement or progress of the Work by any cause beyond the control of Design-Builder, Design-Builder shall be entitled to an equitable extension of the Guaranteed Completion Date for the Contract Milestones, including Substantial Completion of the Work, and/or Final Completion, as applicable. Examples of causes beyond the control of Design-Builder include, but are not limited to, the following to the extent one or a combination of the following are the cause of the applicable delay: (a) acts or omissions of Owner or Others that are not immediately cured upon Design-Builder's written notice to Owner; (b) changes in the Work or the sequencing of the Work ordered by Owner, or arising from decisions of Owner that impact the time of performance of the Work if Contractor notifies Owner immediately and is otherwise entitled to a Change; (c) encountering Hazardous Materials unanticipated by Design-Builder, or concealed or unknown conditions; (d) delay authorized by Owner pending dispute resolution or suspension by Owner under §12.1; (e) transportation delays not reasonably preventable with diligence, given the known supply chain issues; (f) labor disputes not involving Design-Builder; (g) general labor disputes impacting the Project but not specifically related to the Worksite; (h) fire; (i) Terrorism; (j) epidemics, to the extent the impact is not reasonably preventable with diligence, given the knowledge of COVID 19; (k) adverse governmental actions; (l) adverse weather conditions that could not be reasonably anticipated; (m) unusual delay in deliveries of materials or equipment not caused by Design-Builder and not reasonably preventable with diligence, given current market conditions ; and (n) work force unavailability due to illness reasonably causing Design-Builder to restrict the progress of the Work. Design-Builder shall process any requests for equitable extensions of the Contract Milestones in accordance with the provisions of ARTICLE 9.

6.3.2 In addition, if Design-Builder incurs additional costs as a result of a delay that is caused by items (a) through (d) immediately above, Design-Builder shall be entitled to an equitable adjustment in the GMP subject to §6.5; otherwise an extension of time shall be Design-Builder's sole and exclusive relief.

6.3.3 If delays to the Project are encountered for any reason, the Parties agree to take reasonable steps to mitigate the effect of such delays.

6.4 LIQUIDATED DAMAGES

6.4.1 The Parties agree that this Agreement \boxtimes shall provide for the imposition of liquidated damages based on the Guaranteed Completion Date of the respective Contract Milestone.

6.4.1.1 Design-Builder understands that if the Guaranteed Completion Date, as may be amended by subsequent Change Order, is not attained for each Contract Milestone, Owner will suffer damages which are difficult to determine and accurately specify. Design-Builder agrees that if the Guaranteed Completion Date for each Contract Milestone is not attained, Design-Builder shall pay Owner the amount set forth in Section 6.1 as liquidated damages and not as a penalty for each Day that completion extends beyond the respective Guaranteed Completion Date. The liquidated damages provided herein shall be in lieu of all liability for any and all extra costs, losses, expenses, claims, penalties, and any other damages of whatsoever nature incurred by Owner which are occasioned by any delay in achieving any of the Contract Milestones.

6.5 LIMITED MUTUAL WAIVER OF CONSEQUENTIAL DAMAGES Except for damages mutually agreed upon by the Parties as liquidated damages in §6.4, the Parties agree to waive all claims against each other for any consequential damages that may arise out of or relate to this Agreement, except for those specific items of damages excluded from this waiver as mutually agreed upon by the Parties and identified below. Owner agrees to waive damages including but not limited to Owner's loss of use of the Project, any rental expenses incurred, loss of income, profit, or financing related to the Project, as well as

the loss of business, loss of financing, principal office overhead and expenses, loss of profits not related to this Project, loss of reputation, or insolvency. Design-Builder agrees to waive damages including but not limited to loss of business, loss of financing, loss of profits not related to this Project, loss of bonding capacity, loss of reputation, or insolvency. The provisions of this section shall also apply to the termination of this Agreement and shall survive such termination. The following items of damages are excluded from this mutual waiver: N/A

6.5.1 The Parties shall require similar waivers in contracts with Subcontractors and Others retained for the Project.

6.6 Substantial Completion Process

6.6.1 When the Design-Builder considers that the Work, a portion thereof which the Owner agrees, in writing, to accept separately, or a Contract Milestone, is substantially complete, the Design-Builder shall prepare and submit to the Owner a comprehensive list of items to be completed or corrected prior to final payment. Failure to include an item on such list does not alter the responsibility of the Design-Builder to complete all Work in accordance with the Contract Documents.

6.6.2 Upon receipt of the Design-Builder's list, the Owner will make an inspection to determine whether the Work or designated portion thereof is Substantially Complete. If the Owner's inspection discloses any item, whether or not included on the Design-Builder's list, which is not sufficiently complete in accordance with the Contract Documents so that the Owner can occupy or utilize the Work or designated portion thereof for its intended use, the Design-Builder shall complete or correct such item upon notification by the Owner. In such case, the Design-Builder shall then submit a request for another inspection by the Owner to determine Substantial Completion.

6.6.3 When the Work or designated portion thereof is substantially complete, the Owner will prepare a Certificate of Substantial Completion for each Contract Milestone that shall establish the date of Substantial Completion; establish responsibilities of the Owner and Design-Builder for security, maintenance, heat, utilities, damage to the Work and insurance; and fix the time within which the Design-Builder shall finish all items on the list accompanying the Certificate.

6.6.4 The Certificate of Substantial Completion shall be submitted to Design-Builder for their written acceptance of responsibilities assigned to them in the Certificate. Upon such acceptance, and consent of surety if any, the Owner shall make payment associated with the respective Contract Milestone, if any, in Design-Builder's subsequent application for payment. Such payment shall be adjusted for Work that is incomplete or not in accordance with the requirements of the Contract Documents.

ARTICLE 7 PAYMENT

7.1 DESIGN PHASE PAYMENTS

7.1.1 To the extent required by Laws, the cost of services performed directly by Design-Professional is computed separately and is independent from Design-Builder's compensation for work or services performed directly by Design-Builder; these costs shall be shown as separate items on applications for payment and incorporated into the Schedule of Values. If any Design-Professional is retained by Design-Builder, the payments to Design-Professional shall be as detailed in a separate agreement between Design-Builder and Design-Professional.

7.1.2 Owner shall pay Design-Builder for services performed during the Design Phase, as set forth in Exhibit H to Exhibit 1

7.1.3 Compensation for Design Phase services, as part of the Work, shall include Design-Builder's Fee, paid in proportion to the services performed, subject to adjustment

7.1.4 Compensation for Design Phase services, shall be billed in the same manner as Construction Phase payments pursuant to ARTICLE 10.

7.2 PAYMENT TO DESIGN-BUILDER

7.2.1 Owner shall pay Design-Builder for Work performed following the commencement of the Construction Phase on the following basis:

7.2.1.1 the Cost of the Work as allowed in ARTICLE 8;

7.2.1.2 Design-Builder's Fee paid in proportion to the services performed subject to adjustment; and

7.2.1.3 Any shared savings due to Design-Builder pursuant to Section 7.5.

7.2.2 The compensation to be paid by Owner shall be limited to the GMP, which may be adjusted under ARTICLE 9.

7.2.3 However, payment for Construction Phase services shall be computed utilizing the Schedule of Values fully allocating the GMP based on the actual progress of the Work, followed by a reconciliation of the actual Cost of the Work and associated fee, which is due from Design-Builder prior to its Final Application for Payment as further set forth in ARTICLE 10.

7.2.3.1 Schedule of Values. The schedule of values attached hereto as **Exhibit H to Exhibit 1** ("Schedule of Values" or "SOV"), allocates the entire Guaranteed Maximum Price to the various portions of the Work and includes a \$[***] line item (line item 60) for Design-Builder's projected Fee, a \$[***] of which shall not be due until Design-Builder achieves Final Completion. Design-Builder represents that the Schedule of Values allocates the GMP in a manner that can be reasonably supported by the data for expected expenses of the line items included therein. This SOV, shall be used as a basis for reviewing the Design-Builder's Applications for Payment. Any changes to the SOV shall be submitted to the Owner and supported by such data to substantiate its accuracy as the Owner may require, and must be approved by the Owner in writing before it can be used as a basis for reviewing the Design-Builder's subsequent Applications for Payment.

7.3 DESIGN-BUILDER'S FEE Design-Builder's Fee shall be as follows, subject to adjustment as provided in §7.4: [***]% of the Cost of the Work as calculated in **Exhibit 1**, based on the actual Costs of the Work, excluding Design-Builder's Contingency

7.4 ADJUSTMENT IN DESIGN-BUILDER'S FEE Adjustment in Design-Builder's Fee shall be made as follows:

7.4.1 for changes in the Work as provided in ARTICLE 9, Design-Builder's Fee shall be adjusted as follows: [***]% of the Cost of the Work

7.4.2 if Design Builder is placed in charge of managing the replacement of an insured or uninsured loss, Design Builder shall be paid an additional fee in the same proportion that Design Builder's Fee bears to the estimated Cost of the Work for the replacement.

7.5 SHARED SAVINGS If at Final Completion, Cost of the Work plus Design-Builder's Fee is less than (the GMP reduced by \$[***] for Design-Builder's Contingency built into the GMP), the difference shall be shared as follows: The savings, if any, shall accrue seventy-five percent (75%) to Owner and twenty-five percent (25%) to Design-Builder. The shared savings shall be calculated and paid as part of final payment. Design-Builder shall not be entitled to any portion of the shared savings if either Party terminates this Agreement except for a termination for convenience pursuant to \$12.4 or Design-Builder's termination pursuant to \$12.5.

ARTICLE 8 COST OF THE WORK

Owner agrees to pay Design-Builder for the Cost of the Work as defined in this article. This payment shall be in addition to Design-Builder's Fee stipulated in §7.3.

8.1 COST ITEMS FOR DESIGN PHASE SERVICES

8.1.1 Compensation for Design Phase services as provided in §7.1.

8.2 COST ITEMS FOR CONSTRUCTION PHASE SERVICES

8.2.1 Labor Wages directly employed by Design-Builder performing the Work;

8.2.2 Salaries of Design-Builder's employees when stationed at the field office to the extent necessary to complete the applicable Work, employees engaged on the road expediting the production or transportation of material and equipment, and supervisory employees from the principal or branch office as mutually agreed by the Parties in writing;

8.2.3 Cost of employee benefits and taxes including but not limited to workers' compensation, unemployment compensation, social security, health, welfare, retirement, and other fringe benefits as required by law, labor agreements, or paid under Design-Builder's standard personnel policy, insofar as such costs are paid to employees of Design-Builder who are included in the Cost of the Work under §8.2.1 and §8.2.2;

8.2.4 Reasonable transportation, travel, hotel, and moving expenses of Design-Builder's personnel incurred in connection with the Work; fixed hourly rates for the Design-Builder's employees shall be established and included with the GMP Amendment, as mutually agreed upon by the Owner and the Design-Builder. These rates shall include all salaries, employee benefits and taxes, fringe benefits, cell phone allowance, cost of company vehicle (or automobile allowance, as applicable), and the cost of fuel reimbursement for each employee's travel expense.

8.2.5 Cost of all materials, supplies, and equipment incorporated in the Work, including costs of inspection and testing if not provided by Owner, transportation, storage, and handling;

8.2.6 Payments made by Design-Builder to Subcontractors for performed Work;

8.2.7 Fees and expenses for design services procured or furnished by Design-Builder except as provided by the Design-Professional and compensated in §7.1;

8.2.8 Cost, including transportation and maintenance of all materials, supplies, equipment, temporary facilities and hand tools not owned by the workers that are used or consumed in the performance of the Work, less salvage value or residual value; and cost less salvage value on such items used, but not consumed, that remain the property of Design-Builder;

8.2.9 Rental charges of all necessary machinery and equipment, exclusive of hand tools owned by workers, used at the Worksite, whether rented from Design-Builder or others, including installation, repair, and replacement, dismantling, removal, maintenance, transportation, and delivery costs. Rental from unrelated third parties shall be reimbursed at actual cost. Rentals from Design-Builder or its affiliates, subsidiaries, or related parties shall be reimbursed at the prevailing rates in the locality of the Worksite up to eighty-five percent (85%) of the value of the piece of equipment.

8.2.10 Cost of the premiums for all insurance and surety bonds which Design-Builder is required to procure or deems necessary, and approved by Owner, including any additional premium incurred as a result of any increase in the GMP. The Design-Builder's fixed rate for commercial general liability insurance shall be 0.75% of the Cost of the Work.

8.2.11 Sales, use, gross receipts, or other taxes, tariffs, or duties related to the Work for which Design-Builder is liable.

8.2.12 Permits, fees, licenses, tests, and royalties.

8.2.13 Losses, expenses, or damages to the extent not compensated by insurance or otherwise, and the cost of corrective work or redesign during the Construction Phase and for a one-year period following the Date of Substantial Completion, provided that such corrective work or redesign did not arise from Design-Builder's negligence.

8.2.14 All costs associated with establishing, equipping, operating, maintaining, and demobilizing the field office.

8.2.15 All water, power, and fuel costs necessary for the Work.

8.2.16 Cost of removal of all non-hazardous substances, debris, and waste materials.

8.2.17 All costs directly incurred in the performance of the Work or in connection with the Project, and not included in Design-Builder's Fee as set forth in ARTICLE 7, which are reasonably inferable from the Contract Documents.

8.3 DISCOUNTS All discounts for prompt payment shall accrue to Owner to the extent such payments are made directly by Owner. To the extent payments are made with funds of Design-Builder, all cash discounts shall accrue to Design-Builder. All trade discounts, rebates, and refunds, and all returns from sale of surplus materials and equipment, shall be credited to the Cost of the Work.

ARTICLE 9 CHANGES IN THE WORK

Changes in the Work which are within the general scope of this Agreement may be accomplished, without invalidating this Agreement, by Change Order, Interim Directive, or a minor change in the work, subject to the limitations stated in the Contract Documents.

9.1 CHANGE ORDER

9.1.1 Design-Builder may request or Owner, without invalidating this Agreement, may order changes in the Work within the general scope of the Contract Documents consisting of additions, deletions, or other revisions to the GMP or the estimated cost of the work, compensation for Design Phase services, Design-Builder's Fee, or the Guaranteed Completion Date for the Contract Milestones being adjusted accordingly ("Changes"). All such Changes i shall be authorized by applicable Change Order, and processed in accordance with this article.

9.1.2 Each adjustment in the GMP or estimated Cost of the Work resulting from a Change Order shall clearly separate the amount attributable to compensation for Design Phase services, other Cost of the Work, and Design-Builder's Fee, with Design-Builder's Fee not to exceed [***] percent ([***]%).

9.1.3 The Parties shall negotiate an appropriate adjustment to the GMP or the estimated Cost of the Work, compensation for Design Phase services, Design-Builder's Fee, or the Contract Milestones in good faith and conclude negotiations as expeditiously as possible. Acceptance of the Change Order and any adjustment in the GMP, the estimated Cost of the Work, compensation for Design Phase services, Design-Builder's Fee, or the Contract Milestones shall not be unreasonably withheld.

9.1.4 NO OBLIGATION TO PERFORM Design-Builder shall not be obligated to perform changes in the Work until a Change Order has been executed or Interim Directive has been issued.

9.2 INTERIM DIRECTIVE

9.2.1 Owner may issue an Interim Directive directing a change in the Work before agreeing on an adjustment, if any, in the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contract Milestones, and if appropriate, the compensation for Design Phase services or directing Design-Builder to perform Work that Owner believes is not a change. If the Parties disagree that the Interim Directed work is within the scope of the Work, Design-Builder shall perform the disputed Work and furnish Owner with an estimate of the costs to perform the disputed work in accordance with Owner's interpretations.

9.2.2 The Parties shall negotiate expeditiously and in good faith for appropriate adjustments, as applicable, to the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contract Milestones, and if appropriate the compensation for Design Phase services, arising out of the Interim Directive. As the changed Work is performed, Design-Builder shall submit its costs for such Work with its application for payment beginning with the next application for payment within thirty (30) Days of the issuance of the Interim Directive. If there is a dispute as to the cost to Owner, Owner shall pay Design-Builder fifty percent (50%) of its actual (incurred or committed) cost to perform such Work. In such event, the Parties reserve their rights as to the disputed amount, subject to the requirements of ARTICLE 12. Owner's payment does not prejudice its right to be reimbursed should it be determined that the disputed work was within the scope of the Work. Design-Builder's receipt of payment for the disputed work should it be determined that the disputed in applications for payment and shall be paid by Owner in accordance with this Agreement.

9.2.3 When the Parties agree upon the adjustments in the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contract Milestones, and if appropriate the compensation for Design Phase services, for a change in the Work directed by an Interim Directives, such agreement shall be the subject of an appropriate Change Order. The Change Order shall include all outstanding Interim Directives on which Owner and Construction Manager have reached agreement on the GMP or the Contract Milestones issued since the last Change Order.

9.3 MINOR CHANGES IN THE WORK

9.3.1 Design-Builder may make minor changes in the design and construction of the Project consistent with the intent of the Contract Documents which do not involve an adjustment in the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contract Milestones, and do not materially and adversely affect the design of the Project, the quality of any of the materials or equipment specified in the Contract Documents, the performance of any materials, equipment, or systems specified in the Contract Documents, or the quality of workmanship required by the Contract Documents.

9.3.2 Design-Builder shall promptly inform Owner in writing of any such changes and shall record such changes on the Design-Build Documents maintained by Design-Builder.

9.4 CONCEALED OR UNKNOWN SITE CONDITIONS Design Builder is entitled to an adjustment in the GMP or the dates of Substantial and Final Completion to the extent it is entitled to such relief under the terms of the LOI; otherwise, as stated in Section 3.2.2 site conditions will not be the basis for Design-Builder to assert a Change.

9.5 DETERMINATION OF COST

9.5.1 An increase or decrease in the GMP or estimated Cost of the Work resulting from a change in the Work shall be determined by one or more of the following methods:

- 9.5.1.1 unit prices set forth in this Agreement or as subsequently agreed;
- 9.5.1.2 a mutually accepted, itemized lump sum;

9.5.1.3 costs determined as defined in §7.2 and ARTICLE 8 and a mutually acceptable Design-Builder's Fee as determined in §7.4.1; or

9.5.2 If an increase or decrease in GMP or Contract Milestones cannot be agreed to as set forth in §9.5.1 above, and Owner issues an Interim Directive, the cost of the change in the Work shall be determined by the Cost of the Work expense incurred and savings as defined in ARTICLE 8 realized in the performance of the Work resulting from the change. If there is a net increase in the GMP, Design-Builder's Fee shall be adjusted as set forth in §7.4.1. In case of a net decrease in the GMP, Design-Builder's Fee shall be adjusted as forth in §7.4.1. In case of a net decrease in the GMP, Design-Builder's Fee shall not be adjusted unless ten percent (10%) or more of the Project is deleted. Design-Builder shall maintain a documented, itemized accounting evidencing the Cost of Work expenses and savings.

9.5.3 If unit prices are indicated in the Contract Documents or are subsequently agreed to by the Parties, but the character or quantity of such unit items as originally contemplated is so different in a proposed Change Order that the original unit prices will cause substantial inequity to Owner or Design-Builder, such unit prices shall be equitably adjusted.

9.5.4 If Owner and Design-Builder disagree as to whether work required by Owner is within the scope of the Work, Design-Builder shall furnish Owner with an estimate of the costs to perform the disputed work in accordance with Owner's interpretations. If Owner issues a written order for Design-Builder to proceed, Design-Builder shall perform the disputed work and Owner shall pay Design-Builder fifty percent (50%) of its actual, direct cost to perform the work. In such event, both Parties reserve their rights as to whether the work was within the scope of the Work. Owner's payment does not prejudice its right to be reimbursed should it be determined that the disputed work was within the scope of Work. Design-Builder's receipt of payment for the disputed work does not prejudice its right to receive full payment for the disputed work should it be determined that the disputed work work should it be determined that the disputed work should it be

9.6 CLAIMS FOR ADDITIONAL COST OR TIME For any claim for an increase in the GMP, estimated Cost of the Work, Design-Builder's Fee, and the Contract Milestones, and if appropriate the compensation for Design Phase services, Design-Builder shall give Owner written notice of the claim within fourteen (14) Days after the occurrence giving rise to the claim or fourteen (14) Days after Design-Builder first recognizes the condition giving rise to the claim, whichever is later. Except in an emergency, notice shall be given before proceeding with the Work. Claims for design and estimating costs incurred in connection with possible changes requested by Owner, but which do not proceed, shall be made within twenty-one (21) Days after the decision is made not to proceed. Thereafter, Design-Builder shall submit written documentation of its claim, including appropriate supporting documentation, within twenty-one (21) Days after giving notice, unless the Parties mutually agree upon a longer period of time. Owner shall respond in writing denying or approving Design-Builder's claim no later than fourteen (14) Days after receipt of Design-Builder's documentation of claim. Owner's failure to so respond shall be deemed a denial of Design-Builder's claim. Any change in the GMP, estimated Cost of the Work, Design-Builder's Fee, the Contact Milestones, and if appropriate the compensation for Design Phase services, resulting from such claim shall be authorized by Change Order.

9.7 CHANGES NOTICE Owner may direct Design-Builder to perform incidental changes in the Work upon concurrence with Design-Builder that such changes do not involve adjustments in the Cost of the Work or Contract Milestones. Incidental changes shall be consistent with the scope and intent of the Contract Documents. Owner shall initiate an incidental change in the Work by issuing an Interim Directive

9.8 NO WAIVER No action, conduct, omission, prior failure, or course of dealing by the Owner shall act to waive, modify, change or alter the requirement that Change Orders and Construction Change Directives must be in writing and signed as provided herein. Change Orders and Construction Directives are (a) the exclusive methods for effecting any change in the Work, except minor changes as set forth in paragraph 7.4, and (b) except as otherwise expressly provided in the Contract Documents, the exclusive methods for effecting any change in the Work cannot be authorized and, except as expressly provided to the contract Milestones. The Design-Builder understands and agrees that changes in the Work cannot be authorized and, except as expressly provided to the contrary in the Contract Documents, the Guaranteed Maximum Price or Contract Milestones cannot be changed by, implication, oral agreements, actions, inactions, course of conduct, or constructive change order. Any work performed by the Design-Builder and not included in the Contract Documents as provided herein shall be undertaken by the Design-Builder at its own risk and own expense and the Owner shall not be obligated to reimburse or pay the Design-Builder for same. Likewise, Design-Builder shall have no obligation to proceed with any change in the Work unless Design-Builder has received a signed Change Order or Construction Directive as provided herein.

9.9 EXCLUSIVE RELIEF Design-Builder's only relief for a change shall be through the submission of a change order requests in accordance with Section 9.6. Design-Builder shall include and set forth within each change order request, or in a supplement as soon thereafter as practical, a clear and precise description of the changes, the breakdowns of labor and materials for all trades impacted or involved by the changes, the estimated impact on the Construction Schedule, and Design-Builder's efforts to mitigate the impact with the request. To the extent Design-Builder does not timely submit such a change order request in compliance with this provision and Article 7 generally, it is waiving its right to recover any adjustment to the Guaranteed Maximum Price and any adjustment to the Contract Milestones.

9.10 EXECUTION OF CHANGE ORDERS In signing a Change Order, the Owner and Design-Builder acknowledge and agree that (1) the stipulated compensation (adjustment to the Guaranteed Maximum Price or Contract Milestones, or both) set forth in the Change Order includes payment for: (a) the Cost of the Work in any way impacted by the Change Order, (b) Design-Builder's fee for overhead and profit, (c) interruption of Project or Construction Schedules, (d) delay and impact, including cumulative impact, on other Work under the Contract Documents, and (e) extended home office and jobsite overhead, (2) the Change Order constitutes full mutual accord and satisfaction for the change to the Work, (3) Except where expressly stated on the Change Order, Design-Builder is not reserving any rights to pursue subsequent claims related in any way to the Change Order, and (4) No subsequent claim will arise out of or as a result of the signed Change Order, excepting claims arising from Owner's failure to comply with its obligations under the Contract Documents.

ARTICLE 10 PAYMENT FOR CONSTRUCTION PHASE SERVICES

10.1 PROGRESS PAYMENTS

10.1.1 On the 10th Day of each month, Design-Builder shall submit to the Owner an itemized Application for Payment prepared in accordance with the Schedule of Values based on the actual progress of the completed portions of the Work during the prior month. Design-Builder shall utilize the G702/703 or similar form. The application shall be notarized, if required, and supported by all data substantiating the Design-Builder's right to payment that the Owner requires, such as copies of requisitions, and releases and waivers of liens from Subcontractors and suppliers. At minimum, as a condition precedent to payment, Design-Builder shall include (1) a Conditional Interim Waiver in the form attached as **Exhibit 3-1** with respect to the current application and an Unconditional Interim Waiver in the form attached as **Exhibit 3-2** with respect to the previously application for payment from (i) Design-Builder and (ii) any sub-tier entity performing work in excess of \$25,000.00 on the Project (2) a certified list of all sub-tier entities performing work in excess of \$25,000.00, the total current amount of the underlying under which they are performing, the amount previously paid, and the amount currently due to each.

10.1.1.1 Applications for Payment shall not include requests for payment for portions of the Work for which the Design-Builder does not intend to pay a Subcontractor or supplier, unless such Work has been performed by others whom the Design-Builder intends to pay.

10.1.1.2 Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. If approved in advance by the Owner, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Design-Builder with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect the Owner's interest, and shall include the costs of applicable insurance, storage, and transportation to the site, for such materials and equipment stored off the site.

10.1.1.3 Applications for Payment shall show the percentage of completion of each portion of the Work as of the end of the period covered by the Application for Payment. The percentage of completion shall be the lesser of (1) the percentage of that portion of the Work which has actually been completed; or (2) the percentage obtained by dividing (a) the expense that has actually been incurred by the Design-Builder on account of that portion of the Work and for which the Design-Builder has made payment or intends to make payment prior to the next Application for Payment, by (b) the share of the Guaranteed Maximum Price allocated to that portion of the Work in the schedule of values.

10.1.2 Within thirty (30) Days after receipt of each monthly application for payment, Owner shall give written notice to Design-Builder of Owner's acceptance or rejection, in whole or in part, of such application for payment. Within thirty (30) Days after accepting such application, Owner shall pay directly to Design-Builder the appropriate amount for which application for payment is made, less amounts previously paid by Owner. If such application is rejected in whole or in part, Owner shall indicate the reasons for its rejection. If Owner and Design-Builder cannot agree on a revised amount then, within thirty (30) Days after its initial rejection in part of such application, Owner shall pay directly to Design-Builder the appropriate amount for which application for payment is made, less amounts previously paid by Owner. Those items not rejected by Owner for which application for payment is made, less amounts previously paid by Owner. Those items rejected by Owner shall be due and payable when the reasons for the rejection have been removed.

10.1.3 If for any reason not the fault of Design-Builder, Design-Builder does not receive a progress payment from Owner within thirty (30) Days after the time such undisputed payment is due, then Design-Builder, upon giving thirty (30) Days' written notice to Owner, and without prejudice to and in addition to any other legal remedies, may stop Work until payment of the full undisputed amount owing to Design-Builder has been received, including interest for late payment. If Design-Builder incurs costs or is delayed resulting from shutdown, delay, and start-up, Design-Builder may seek an equitable adjustment in compensation, time, or both, under ARTICLE 9.

10.1.4 Payments due but unpaid pursuant to §10.1.2 may bear interest from the date payment is due at the prime rate prevailing at the place of the Project.

10.1.5 Design-Builder warrants and guarantees that title to all Work, materials, and equipment covered by an application for payment, whether incorporated in the Project or not, will pass to Owner upon receipt of such payment by Design-Builder, free and clear of all liens, claims, security interests, or encumbrances, hereinafter referred to as liens.

10.1.6 Owner's progress payment, occupancy, or use of the Project, whether in whole or in part, shall not be deemed an acceptance of any Work not conforming to the requirements of the Contract Documents.

10.1.7 Upon Substantial Completion of the Work, Owner shall pay Design-Builder the unpaid balance of the Cost of the Work, compensation for Design Phase services, and Design-Builder's Fee, less one hundred and fifty percent (150%) of the cost of completing any unfinished items as agreed to between Owner and Design-Builder as to extent and time for completion. Owner thereafter shall pay Design-Builder monthly the amount retained for unfinished items as each item is completed.

10.1.8 STORED MATERIALS AND EQUIPMENT Unless otherwise provided in the Contract Documents, applications for payment may include materials and equipment not yet incorporated into the Work but delivered to and suitably stored onsite or offsite, including applicable insurance, storage, and costs incurred transporting the materials to an offsite storage facility. Approval of payment applications for stored materials and equipment stored offsite shall be conditioned on submission by Design-Builder of bills of sale and proof of required insurance, or such other procedures satisfactory to Owner to establish the proper valuation of the stored materials and equipment, Owner's title to such materials and equipment, and to otherwise protect Owner's interests therein, including transportation to the worksite.

10.1.9 RETAINAGE Owner and Design-builder have agreed to five percent (5%) retainage and to deposit the retainage in an interest-bearing escrow account in compliance with T.C.A. § 66-34-101. *et. seq.* Owner and Contractor have entered into the Retainage Escrow Agreement with Regions Bank to hold the retainage in compliance with T.C.A. § 66-34-101. *et. seq.* There shall be no retention on Design-Builder's fee and Design-Builder's staffing/general conditions.

10.2 ADJUSTMENT OF DESIGN-BUILDER'S APPLICATION FOR PAYMENT Owner may adjust or reject an application for payment or nullify a previously approved Design-Builder application for payment, in whole or in part, as may reasonably be necessary to protect Owner from loss or damage based upon the following, to the extent that Design-Builder is responsible under this Agreement:

10.2.1 Design-Builder's repeated failure to perform the Work as required by the Contract Documents;

10.2.2 except as accepted by the insurer providing Builder's Risk or other property insurance covering the project, loss or damage arising out of or relating to this Agreement and caused by Design-Builder to Owner or others to whom Owner may be liable;

10.2.3 Design-Builder's failure to properly pay either Design-Professional, Subcontractors, or Suppliers for labor, materials, equipment, or supplies furnished in connection with the Work, provided that Owner is making payments to Design-Builder in accordance with this Agreement;

10.2.4 Rejected or Defective Work not corrected in a timely fashion;

10.2.5 reasonable evidence of delay in performance of the Work such that the Work will not be completed by the Contract Milestones, and that the unpaid balance of the GMP is not sufficient to offset any direct damages that may be sustained by Owner as a result of the anticipated delay caused by Design-Builder;

10.2.6 reasonable evidence demonstrating that the unpaid balance of the GMP is insufficient to fund the cost to complete the Work; and

10.2.7 uninsured third-party claims involving Design-Builder or reasonable evidence demonstrating that third-party claims are likely to be filed unless and until Design-Builder furnishes Owner with adequate security in the form of a surety bond, letter of credit, or other collateral or commitment sufficient to discharge such claims if established. In the case of a lien, Design-Builder shall furnish Owner with a recorded lien release bond, completely releasing the property, with the exception of liens filed due Owner's failure to pay in accordance with the Agreement.

10.2.8 If Owner intends to disapprove, withhold, or otherwise nullify all or part of a payment otherwise due, Owner shall give written notice to Design-Builder, before such payment is due, stating its specific reasons for such disapproval or nullification, and the remedial actions to be taken by Design-Builder in order to receive payment. When the above reasons for disapproving or nullifying an application for payment are removed, payment will be promptly made for the amount previously withheld.

10.3 OWNER OCCUPANCY OR USE OF COMPLETED OR PARTIALLY COMPLETED WORK Owner may use completed or partially completed portions of the Work when (a) the portion of the Work is designated in a Certificate of Substantial Completion (or earlier with Contractor's permission, which shall be reasonably provided), (b) appropriate insurer(s) or sureties consent to the occupancy or use, and (c) appropriate public authorities authorize the occupancy or use. The Parties shall agree whether such partial occupancy or use shall constitute Substantial Completion of that portion of the Work, as according to the Owner and Design-Builder's mutually agreed terms. Design-Builder shall not unreasonably withhold consent to partial occupancy or use, provided such partial occupancy or use is of value to Owner.

10.4 FINAL PAYMENT

10.4.1 Final Payment, consisting of the unpaid balance of the Cost of the Work, compensation for Design Phase services, and Design-Builder's Fee, shall be due and payable when the work is fully completed, all punch list work has been completed, and Design-Builder has otherwise achieved Final Completion and all conditions of this Section 10.4 have been satisfied by Design-Builder. Before issuance of final payment, Owner may request satisfactory evidence that all payrolls, material bills, and other indebtedness connected with the Work have been paid or otherwise satisfied.

10.4.2 Design-Builder's Cost Reconciliation

10.4.2.1 Prior to submitting an application for Final Payment, Design-Builder shall submit to Owner a final accounting, including an itemized list of the actual Cost of the Work, along with supporting documentation.

Within 15 days of the Owner's receipt of the Design-Builder's final accounting for the Cost of the Work, the Owner shall conduct an audit of the Cost of the Work or notify the Design-Builder that it will not conduct an audit. Owner's audit shall be conducted and completed within 45 days of Design-Builder's final accounting for the Cost of the Work.

10.4.2.2 If the Owner conducts an audit of the Cost of the Work, the Owner shall, within 10 days after completion of the audit, submit a written report based upon the auditors' findings to the Design-Builder. If the auditors' report concludes that the Cost of the Work, as substantiated by the Design-Builder's final accounting, is less than claimed by the Design-Builder, the Design-Builder shall be entitled to request mediation of the disputed amount with the auditor. Failure to request mediation within 10-days shall result in the substantiated amount reported by the auditor becoming binding on the Design-Builder as the final Cost of the Work.

10.4.2.3 The final amount determined by the auditor, subject to Design-Builder's right to commence arbitration, shall be the final Cost of the Work, and the basis for the calculation of any savings due to Design-Builder pursuant to Section 7.5.

10.4.3 Design-Builder's Application for Final Payment

10.4.3.1 Design-Builder shall submit to the Owner a final Application for Payment prepared in accordance with the Schedule of Values, and based on the final Cost of the Work as agreed, or otherwise determined by the auditor. The application shall be notarized, if required, and supported by all data substantiating the Design-Builder's right to payment that the Owner requires, such as copies of requisitions, and releases and waivers of liens from

Subcontractors and suppliers. At minimum, as a condition precedent to payment, Design-Builder shall include (1) a Conditional Final Waiver in the form attached as **Exhibit 3-3** with respect to the current application from (i) Design-Builder and (ii) any sub-tier entity performing work in excess of \$25,000.00 on the Project (2) a certified list of all sub-tier entities performing work in excess of \$25,000.00, the total current amount of the underlying under which they are performing, the amount previously paid, and the final amount due to each. In addition, in exchange for the Final Payment, Design-Builder shall provide an Unconditional Final Waiver in the form attached as **Exhibit 3-4** from (i) Design-Builder and (ii) any sub-tier entity performing work in excess of \$25,000.00 on the Project.

10.4.4 Withholding, Setoff, and Conditions Precedent to Final Payment

10.4.4.1 If Owner disputes, Design-Builder's final application or any portion thereof, or otherwise asserts that it is entitled to a setoff or other payment from Design-Builder Owner shall provide written notification to Design-Builder within fifteen (15) Days following Design-Builder's application for final payment or the event or circumstance giving rise to Owner's claim for setoff or other payment, subject to Design-Builder's delivery to Owner of the following: (1) an affidavit that payrolls, bills for materials and equipment, and other indebtedness connected with the Work for which the Owner or the Owner's property might be responsible or encumbered have, or will be been paid in exchange for final payment, or otherwise satisfied, (2) a certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect, (3) a written statement that the Design-Builder knows of no reason that the insurance will not be renewable to cover the period required by the Contract Documents, (4) the delivery of as-built drawings (5) documentation of any special warranties, such as manufacturers' warranties or specific Subcontractor warranties, and (6) a notice of completion in conformance with TCA 66-11-143 recorded more than 10 days, prior to the issuance of final payment.

10.4.5 In making final payment Owner waives all claims except for:

10.4.5.1 outstanding liens;

10.4.5.2 improper workmanship or defective materials appearing within one year after the Date of Substantial Completion;

10.4.5.3 work not in conformance with the Contract Documents; and

10.4.5.4 terms of any special warranties required by the Contract Documents.

10.4.6 In accepting final payment, Design-Builder waives all claims except those previously made in writing and which have been submitted to mediation or arbitration in accordance with Article 13 and which remain unsettled.

ARTICLE 11 INDEMNITY, INSURANCE, AND BONDS

11.1 INDEMNITY

11.1.1 To the fullest extent permitted by law, Design-Builder shall indemnify and hold harmless Owner, Owner's officers, directors, members, and employees (the Indemnitees) from all claims for bodily injury and property damage other than to the Work itself and other property required to be insured, including reasonable attorneys' fees, costs and expenses, that may arise from the performance of the Work, but only to the extent caused by the negligent or intentionally wrongful acts or omissions of Design-Builder, Subcontractors, or anyone employed directly or indirectly by

any of them or by anyone for whose acts any of them may be liable. Design-Builder shall not be required to indemnify or hold harmless the Indemnitees for any negligent or intentionally wrongful acts or omissions of the Indemnitees. Design-Builder shall be entitled to reimbursement of any defense costs paid above Design-Builder's percentage of liability for the underlying claim to the extent provided for by the subsection below.

11.1.2 To the fullest extent permitted by law, Owner shall indemnify and hold harmless Design-Builder, its officers, directors or members, Subcontractors or anyone employed directly or indirectly by any of them or anyone for whose acts any of them may be liable from all claims for bodily injury and property damage, other than property insured, including reasonable attorneys' fees, costs, and expenses, that may arise from the performance of work by Owner or Others, but only to the extent caused by the negligent or intentionally wrongful acts or omissions of Owner or Others. Owner shall be entitled to reimbursement of any defense costs paid above Owner's percentage of liability for the underlying claim to the extent provided for by the subsection above.

11.1.3 NO LIMITATION ON LIABILITY In any and all claims against the Indemnitees by any employee of Design-Builder, anyone directly or indirectly employed by Design-Builder or anyone for whose acts Design-Builder may be liable, the indemnification obligation shall not be limited in any way by any limitation on the amount or type of damages, compensation or benefits payable by or for Design-Builder under workers' compensation acts, disability benefit acts, or other employee benefit acts.

11.2 DESIGN-BUILDER'S LIABILITY INSURANCE

11.2.1 Before commencing the Work and as a condition precedent to payment, Design-Builder shall procure and maintain in force Workers' Compensation Insurance, Employers' Liability Insurance, Business Automobile Liability Insurance, and Commercial General Liability Insurance (CGL). The CGL policy shall include coverage for liability arising from premises, operations, independent contractors, products-completed operations, personal injury and advertising injury, contractual liability, and broad form property damage. Design-Builder shall maintain completed operations liability insurance for one year after Substantial Completion, or as required by the Contract Documents, whichever is longer. Design-Builder's Employers' Liability, Business Automobile Liability, and CGL policies shall be written with at least the following limits of liability:

- 11.2.1.1 Employers' Liability Insurance
 - (a) \$1,000,000 bodily injury by accident per accident
 - (b) \$1,000,000 bodily injury by disease policy limit
 - (c) \$1.000,000 bodily injury by disease per employee
- 11.2.1.2 Business Automobile Liability Insurance per accident \$2,000,000.
- 11.2.1.3 Commercial General Liability Insurance
 - (a) Per occurrence \$2,000,000
 - (b) General aggregate \$4,000,000
 - (c) Products/completed operations aggregate \$4,000,000
 - (d) Personal and advertising injury limit \$2,000,000

11.2.2 Employers' Liability, Business Automobile Liability, and CGL coverage required under §11.2.1 may be arranged under a single policy for the full limits required or by a combination of underlying policies with the balance provided by Excess or Umbrella Liability policies.

11.2.3 Design-Builder shall maintain in effect all insurance coverage required under §11.2.1 with insurance companies lawfully authorized to do business in the jurisdiction in which the Project is located. If Design-Builder fails to obtain or maintain any insurance coverage required under this Agreement, Owner may purchase such coverage and charge the expense to Design-Builder, or terminate this Agreement.

11.2.4 To the extent commercially available to Design-Builder from its current insurance company, insurance policies required under §11.2.1 shall contain a provision that the insurance company or its designee must give Owner written notice transmitted in paper or electronic format: (a) 30 days before coverage is nonrenewed by the insurance company and (b) within 10 business days after cancelation of coverage by the insurance company. Prior to commencing the Work and upon renewal or replacement of the insurance policies, Design-Builder shall furnish Owner with certificates of insurance until one year after Substantial Completion of the Work or longer if required by the Contract Documents. In addition, if any insurance policy required under §11.2.1 is not to be immediately replaced without lapse in coverage when it expires, exhausts its limits, or is to be cancelled, Design-Builder shall give Owner prompt written notice upon actual or constructive knowledge of such condition.

11.3 PROPERTY INSURANCE

11.3.1 Unless otherwise directed in writing by Owner, before starting the Work, Design-Builder shall obtain and maintain a Builder's Risk Policy upon the entire Project for the full cost of replacement at the time of loss, including existing structures. This insurance shall also: (a) name Design-Builder, Subcontractors, Subsubcontractors, Suppliers, and Design Professional as insureds; (b) be written in such form as to cover all risks of physical loss except those specifically excluded by the policy; and (c) insure at least against and not exclude:

11.3.1.1 the perils of fire, lightning, explosion, windstorm, hail, smoke, aircraft (except aircraft, including helicopter, operated by or on behalf of the Design-Builder) and vehicles, riot and civil commotion, theft, vandalism, malicious mischief, debris removal, flood, earthquake, earth movement, water damage, wind damage, testing if applicable, collapse however caused;

11.3.1.2 damage resulting from defective design, workmanship, or material;

11.3.1.3 coverage extension for damage to existing buildings, plant, or other structures at the Worksite, when the Project is contained within or attached to such existing buildings, plant, or structures. Coverage shall be to the extent loss or damage arises out of Constructor's activities or operations at the Project;

11.3.1.4 equipment breakdown, including mechanical breakdown, electrical injury to electrical devices, explosion of steam equipment, and damage to steam equipment caused by a condition within the equipment;

11.3.1.5 testing coverage for running newly installed machinery and equipment at or beyond the specified limits of their capacity to determine whether they are fit for their intended use; and

11.3.1.6 physical loss resulting from Terrorism.

11.3.2 The Party that is the primary cause of a Builder's Risk Policy claim shall be responsible for any deductible amounts or coinsurance payments. If no Party is the primary cause of a claim, then the Party obtaining and maintaining the Builder's Risk Policy pursuant to §11.3.1 shall be responsible for the deductible amounts or coinsurance payments. This policy shall provide for a waiver of subrogation. This insurance shall remain in effect until final payment has been made or until no person or entity other than Owner has an insurable interest in the property to be covered by this insurance, whichever is sooner. Partial occupancy or use of the Work shall not commence until Design-Builder has secured the consent of the insurance company or companies providing the coverage required in this subsection. Before commencing the Work, Design-Builder shall provide a copy of the property policy or policies obtained in compliance with §11.3.1

11.3.3 If Owner elects to purchase the property insurance required by this Agreement, including all of the same coverages and deductibles for the same duration specified in §11.3.1, then Owner shall give written notice to Design-Builder and the Design Professional before the Work is commenced and provide a copy of the property policy or policies obtained in compliance with §11.3.1. Owner may then provide insurance to protect its interests and the interests of the Design-Builder, Subcontractors, Suppliers, and Subsubcontractors. The cost of this insurance shall be paid by Owner in a Change Order. If Owner gives written notice of its intent to purchase property insurance required by this Agreement and fails to purchase or maintain such insurance, Owner shall be responsible for costs reasonably attributed to such failure.

11.3.4 The Parties waive all rights against each other and their respective employees, agents, contractors, subcontractors and subsubcontractors, and design professionals for damages caused by risks covered by the property insurance except such rights as they may have to the proceeds of the insurance.

11.3.5 To the extent of the limits of Design-Builder's CGL specified in §11.2.1, Design-Builder shall indemnify and hold harmless Owner against any and all liability, claims, demands, damages, losses, and expenses, including attorneys' fees, in connection with or arising out of any damage or alleged damage to any of Owner's existing adjacent property that may arise from the performance of the Work, to the extent of the negligent acts or omissions of Design-Builder, Subcontractor, or anyone employed directly or indirectly by any of them or by anyone for whose acts any of them may be liable.

11.3.6 RISK OF LOSS Except to the extent a loss is covered by applicable insurance, risk of loss or damage to the Work shall be upon the Party obtaining and maintaining the Builder's Risk Policy pursuant to section 11.3.1 until the Date of Final Completion.

11.4 ADDITIONAL GENERAL LIABILITY COVERAGE Owner 🗆 shall/ 🗵 shall not require Design-Builder to purchase and maintain additional liability coverage. If required, Design-Builder shall provide:

11.4.1 Additional Insured. Owner shall be named as an additional insured on Design-Builder's CGL specified, for on-going operations and completed operations, excess/umbrella liability, commercial automobile liability, and any required pollution liability, but only with respect to liability for bodily injury, property damage, or personal and advertising injury to the extent caused by the negligent acts or omissions of Design-Builder, or those acting on Design-Builder's behalf, in the performance of Design-Builder's Work for Owner at the Worksite. The insurance of the Design-Builder and its Subcontractors shall be primary and non-contributory to any insurance available to the Additional Insureds.

11.4.2 OCP. Design-Builder shall provide an Owners' and Contractors' Protective Liability Insurance ("OCP") policy with limits equal to the limits on CGL specified, or limits as otherwise required by Owner.

Any documented additional cost in the form of a surcharge associated with procuring the additional general liability coverage in accordance with this subsection shall be paid by Owner directly or the costs may be reimbursed by Owner to Design-Builder and by increasing the GMP to correspond to the actual cost required to purchase and maintain the coverage. Prior to commencement of the Work, Design-Builder shall provide either a copy of the OCP policy, or a certificate and endorsement evidencing that Owner has been named as an additional insured, as applicable. The insurance of the Design-Builder shall be primary and non-contributory to any insurance available to the Additional Insureds

11.5 ROYALTIES, PATENTS, AND COPYRIGHTS Design-Builder shall pay all royalties and license fees which may be due on the inclusion of any patented or copyrighted materials, methods, or systems selected by Design-Builder and incorporated in the Work. Design-Builder shall defend, indemnify, and hold Owner harmless from all suits or claims for infringement of any patent rights or copyrights arising out of such selection. Owner agrees to defend, indemnify and hold Design-Builder harmless from any suits or claims of infringement of any patent rights or copyrights arising out of any patented or copyrighted materials, methods, or systems specified by Owner.

11.6 PROFESSIONAL LIABILITY INSURANCE Design-Builder shall obtain, either itself or through Design Professional, professional liability insurance for claims arising from the negligent performance of professional services under this Agreement, which shall be:

 \boxtimes Practice Policy or \square Project Specific Coverage written for not less than four million dollars (\$4,000,000) per claim and in the aggregate with a deductible not to exceed Twenty-Five Thousand dollars (\$25,000). The Professional Liability Insurance shall include prior acts coverage sufficient to cover all services rendered by the Design-Professional. This coverage shall be continued in effect through the statute of limitations of the State of Tennessee.

11.7 BONDING Performance and Payment Bonds are not required of Design-Builder.

ARTICLE 12 SUSPENSION, NOTICE TO CURE, AND TERMINATION

12.1 SUSPENSION BY OWNER FOR CONVENIENCE

12.1.1 Owner may order Design-Builder in writing to suspend, delay, or interrupt all or any part of the Work without cause for its convenience.

12.1.2 Adjustments caused by suspension, delay, or interruption shall be made for increases in the GMP, compensation for Design Phase services, Design-Builder's Fee, or the Contract Milestones. No adjustment shall be made if Design-Builder is or otherwise would have been responsible for the suspension, delay, or interruption of the Work, or if another provision of this Agreement is applied to render an equitable adjustment.

12.2 NOTICE TO CURE A DEFAULT If Design-Builder persistently fails to supply enough qualified workers, proper materials, or equipment to maintain the approved Schedule of the Work, or fails to make prompt payment to its workers, Subcontractors, or Suppliers, disregards Laws or orders of any public authority having jurisdiction, or is otherwise guilty of a material breach of a provision of this Agreement, Design-Builder may be deemed in default.

12.2.1 If Design-Builder fails within seven (7) Days after receipt of written notice to commence and continue satisfactory correction of such default, then Owner shall give Design-Builder and, if applicable, the surety, a second notice to correct the default within a three (3) Day period.

12.2.2 After receiving Owner's written notice, if Design-Builder fails to promptly commence and continue satisfactory correction of the default, then Owner without prejudice to any other rights or remedies may: (a) take possession of the Worksite, or portion thereof; (b) complete the Work, or portion thereof, utilizing any reasonable means; (c) withhold payment due to Design-Builder; (d) supplement Design/Builder and/or (e) as Owner deems necessary, supply workers and materials, equipment, and other facilities for the satisfactory correction of the default, and charge Design-Builder the costs and expenses, including reasonable Overhead, profit, and attorneys' fees.

12.2.3 In the event of an emergency affecting the safety of persons or property, Owner may immediately commence and continue satisfactory correction of a default without first giving written notice to Design-Builder, but shall give Design-Builder prompt notice.

12.3 OWNER'S RIGHT TO TERMINATE FOR DEFAULT

12.3.1 TERMINATION BY OWNER FOR DEFAULT Upon expiration of the second notice for default period pursuant to §12.2 and absent appropriate corrective action, Owner may terminate this Agreement by written notice. Termination for default is in addition to any other remedies available to Owner under §12.2. If Owner's costs arising out of Design-Builder's failure to cure, including the costs to complete the Work and reasonable attorneys' fees, exceed the GMP, Design-Builder shall be liable to Owner for such excess costs. If Owner's costs are less than the GMP, Owner shall pay the difference to Design-Builder. If Owner exercises its rights under this section, upon the request of Design-Builder, Owner shall furnish to Design-Builder a detailed accounting of the costs incurred by Owner.

12.3.2 If Design-Builder files a petition under the Bankruptcy Code, this Agreement shall terminate if: (a) Design-Builder or Design-Builder's trustee rejects the Agreement; (b) a default occurred and Design-Builder is unable to give adequate assurance of required performance; or (c) Design-Builder is otherwise unable to comply with the requirements for assuming this Agreement under the applicable provisions of the Bankruptcy Code.

12.3.3 Owner shall make reasonable efforts to mitigate damages arising from Design-Builder default, and shall promptly invoice Design-Builder for all amounts due pursuant to §12.2.

12.4 TERMINATION BY OWNER FOR CONVENIENCE If Owner terminates this Agreement other than as set forth in §12.2, Owner shall pay Design-Builder for all Work executed and for all proven loss, cost, or expense in connection with the Work, plus all demobilization costs. In addition, Design-Builder shall be paid an amount calculated as set forth below:

12.4.1 N/A.

12.4.2 If Owner terminates this Agreement after commencement of the Construction Phase, Design-Builder shall be paid for (a) the Construction Phase services provided to date pursuant to §7.2.1; (b) reasonable attorneys' fees and costs related to termination; and (c) a premium as set forth below: 1% of the Work completed at the time of termination.

12.4.3 Owner shall also pay to Design-Builder fair compensation, either by purchase or rental at the election of Owner, for all equipment retained. Owner shall assume and become liable for obligations, commitments, and unsettled claims that Design-Builder has previously undertaken or incurred in good faith in connection with the Work or as a result of the termination of this Agreement. As a condition of receiving the payments provided under this article, Design-Builder shall cooperate with Owner by taking all steps necessary to accomplish the legal assignment of Design-Builder's rights and benefits to Owner, including the execution and delivery of required papers.

12.5 TERMINATION BY DESIGN-BUILDER

12.5.1 Seven (7) Days after Owner's receipt of Design-Builder's written notice, Design-Builder may terminate this Agreement, if the Work has been stopped for a ninety (90) Day period through no fault of Design-Builder the Work is suspended by Owner for Convenience.

12.5.2 In addition, upon seven (7) Days' written notice to Owner and an opportunity to cure within three (3) Days, Design-Builder may terminate this Agreement if Owner:

12.5.2.1 fails to furnish reasonable evidence pursuant to §4.2 that sufficient funds are available and committed for Project financing; or

12.5.2.2 assigns this Agreement over Design-Builder's reasonable objection; or

12.5.2.3 fails to pay Design-Builder any undisputed amount in accordance with this Agreement and Design-Builder stopped Work in compliance with §10.1.3; or

12.5.2.4 otherwise materially breaches this Agreement.

12.5.3 Upon termination by Design-Builder in accordance with this section, Design-Builder is entitled to recover from Owner payment for all Work executed and for all proven loss, cost or expense in connection with the Work, plus all demobilization costs and reasonable damages. In addition, Design-Builder shall be paid an amount calculated as set forth either in §12.4.1 or §12.4.2, depending on when the termination occurs, and §12.4.3.

ARTICLE 13 DISPUTE MITIGATION AND RESOLUTION

13.1 WORK CONTINUANCE AND PAYMENT Unless otherwise agreed in writing, Design-Builder shall continue the Work and maintain the approved schedules during any dispute mitigation or resolution proceedings. If Design-Builder continues to perform, Owner shall continue to make payments in accordance with the Agreement.

13.2 DIRECT SETTLEMENT DISCUSSIONS If the Parties cannot reach resolution on a matter relating to or arising out of this Agreement, the Parties shall endeavor to reach resolution through good faith direct discussions between the Parties' representatives, who shall possess the necessary authority to resolve such matter and who will record the date of first discussions. If the Parties' representatives are not able to resolve such matter within five (5) Business Days from the date of first discussion, the Parties' representatives shall immediately inform senior executives of the Parties in writing that a resolution could not be reached. Upon receipt of such notice, the senior executives of the Parties shall meet within five (5) Business Days to endeavor to reach resolution. If the dispute remains unresolved after fifteen (15) Days from the date of first discussion, the Parties shall submit such matter to the dispute mitigation and dispute resolution procedures selected below.

13.3 OMITTED

13.4 MEDIATION If direct discussions pursuant to §13.2 do not result in resolution of the matter and no dispute mitigation procedure is selected under §13.3, the Parties shall endeavor to resolve the matter by mediation through the current Construction Industry Mediation Rules of the American Arbitration Association (AAA), or the Parties may mutually agree to select another set of mediation rules. The administration of the mediation shall be as mutually agreed by the Parties. The mediation shall be convened within thirty (30) Business Days of the matter first being discussed and shall conclude within forty-five (45) Business Days of the matter first being discussed. Either Party may terminate the mediation at any time after the first session by written notice to the non-terminating Party and mediator. The costs of the mediation shall be shared equally by the Parties. The Parties choose mediation through:

🗵 the current Construction Industry Mediation Rules of the American Arbitration Association (AAA), and administered by AAA.

□ the current Mediation Guidelines of JAMS and administered by JAMS

 \Box the current rules and administration by [____].

If no box is checked the default is AAA rules and administration.

13.5 BINDING DISPUTE RESOLUTION If the matter is unresolved after submission of the matter to a mitigation procedure or to mediation, the Parties shall submit the matter to the binding dispute resolution procedure selected below:

13.5.1 ARBITRATION

☑ The Parties choose binding arbitration for any claim or dispute arising out of or relating to this Agreement. **EACH PARTY WAIVES THEIR RIGHT TO BE HEARD IN A COURT OF LAW**, with or without a jury. Arbitration does not involve a judge or jury. Instead, an arbitrator with the power to award damages and other appropriate relief will decide claims and disputes. An arbitrator's award shall be final and binding upon the Parties, and judgment may be entered upon it in any court having jurisdiction.

13.5.1.1 Neither Party may commence arbitration if the claim or cause of action would be barred by the applicable statute of limitations had the claim or cause of action been filed in a state or federal court. Receipt of a demand for arbitration by the person or entity administering the arbitration shall constitute the commencement of legal proceedings for the purposes of determining whether a claim or cause of action is barred by the applicable statute of limitations. If, however, a state or federal court exercising jurisdiction over a timely filed claim or cause of action orders that the claim or cause of action be submitted to arbitration, the arbitration proceeding shall be deemed commenced as of the date the court action was filed, provided that the Party asserting the claim or cause of action files its demand for arbitration with the person or entity administering the arbitration within thirty (30) Days after the entry of such order.

13.5.1.2 The arbitration shall use the following rules:

☑ the current AAA Construction Industry Arbitration Rules and AAA administration..

□ the current JAMS Engineering and Construction Arbitration Rules and Procedures and administered by JAMS; or

 \Box the current arbitration rules of [____] and administered by [____].

13.5.2 LITIGATION - Omitted

 $13.5.3 \boxtimes$ COSTS The costs of any binding dispute resolution procedures and reasonable attorneys' fees of the prevailing Party shall be borne by the non-prevailing Party, as determined by the adjudicator of the dispute. A party shall not be considered a prevailing Party unless the judgment on the merits is more favorable than the last written settlement offer made to it more than thirty (30) days before the commencement of the arbitration hearing.

13.5.4 VENUE The Project location shall serve as the venue.

13.6 MULTIPARTY PROCEEDING All Parties necessary to resolve a matter agree to be parties to the same dispute resolution proceeding, if possible. Appropriate provisions shall be included in all other contracts relating to the Work to provide for the joinder or consolidation of such dispute resolution proceedings.

13.7 LIEN RIGHTS. Nothing in this article shall limit any rights or remedies not expressly waived by Design-Builder which Design-Builder may have under lien laws.

ARTICLE 14 MISCELLANEOUS PROVISIONS

14.1 EXTENT OF AGREEMENT Except as expressly provided, this Agreement is for the exclusive benefit of the Parties, and not for the benefit of any third party. This Agreement represents the entire and integrated agreement between the Parties, and supersedes all prior negotiations, representations, or agreements, either written or oral. This Agreement and each and every provision is for the exclusive benefit of the Parties and not for the benefit of any third party.

14.2 ASSIGNMENT Except as to the assignment of proceeds, neither Party shall assign its interest in this Agreement without the written consent of the other Party. The terms and conditions of this Agreement shall be binding upon both Parties, their partners, successors, assigns, and legal representatives. Neither Party shall assign the Agreement as a whole without written consent of the other except that Owner may assign the Agreement to a wholly-owned subsidiary of Owner when Owner has fully indemnified Design-Builder or to an institutional lender providing construction financing for the Project as long as the assignment is no less favorable to Design-Builder than this Agreement. In the event of such assignment, Design-Builder shall execute any consent reasonably required. In such event, the wholly-owned subsidiary or lender shall assume Owner's rights and obligations under the Contract Documents. If either Party attempts to make such an assignment, that Party shall nevertheless remain legally responsible for all obligations under the Agreement, unless otherwise agreed by the other Party.

14.3 GOVERNING LAW The law in effect at the location of the Project shall govern this Agreement.

14.4 SEVERABILITY The partial or complete invalidity of any one or more provisions of this Agreement shall not affect the validity or continuing force and effect of any other provision.

14.5 NOTICE Unless changed in writing, a Party's address indicated in Article 1 shall be used when delivering notice to a physical address. Except for Agreement termination and as otherwise specified in the Contract Documents, notice is effective upon transmission by any effective means, including U.S. postal service and overnight delivery service; however, Design-Builder's notice is only effective if it is also emailed to Owner's Representative and Owner's Senior Vice President/General Counsel, Kevin Maxwell at kmaxwell@smith-wesson.com.

14.6 NO WAIVER OF PERFORMANCE The failure of either Party to insist, in any one or more instances, on the performance of any of the terms, covenants, or conditions of this Agreement, or to exercise any of its rights, shall not be construed as a waiver or relinquishment of such term, covenant, condition, or right with respect to further performance.

14.7 TITLES The titles given to the articles and sections are for ease of reference only and shall not be relied upon or cited for any other purpose.

14.8 JOINT DRAFTING The Parties expressly agree that this Agreement was jointly drafted, and that both had opportunity to negotiate its terms and to obtain the assistance of counsel in reviewing its terms prior to execution. Therefore, this Agreement shall be construed neither against nor in favor of either Party, but shall be construed in a neutral manner.

ARTICLE 15 CONTRACT DOCUMENTS

15.1 CONTRACT DOCUMENTS The Contract Documents are as follows:

(a) This Agreement, including the Exhibits listed in 2.4.2.

(b)The Design Development Documents upon Owner approval pursuant to §3.1.6.

(c) The Construction Documents upon Owner approval under §3.1.7.

(d) Change Order, Interim Directives, and amendments issues in accordance with this Agreement.

15.2 ORDER OF PRECEDENCE The documents shall govern in the following order: (a) Change Orders and written amendments to this Agreement, including Amendment 1; (b) this Agreement; (c) design documents approved by Owner pursuant to §3.1.4 through §3.1.7 in order of the most recently approved; (d) information furnished by Owner pursuant to §3.7.4 or designated as a Contract Document in §15.1; (e) other documents listed in this Agreement. Except as otherwise provided, among categories of documents having the same order of precedence, the term or provision that includes the latest date shall control. Where figures are given, they shall be preferred to scaled dimensions. Unless otherwise specifically defined in this Agreement, any terms that have well-known technical or trade meanings shall be interpreted in accordance with their well-known meanings

15.3 CONFLICTS In case of any inconsistency, conflict or ambiguity among the Contract Documents, Design-Builder shall comply with the provision setting forth the higher standard of conduct, or more stringent obligations upon Design-Builder.

OWNER: SMITH & WESSON BRANDS, INC.

BY:	Mark Smith	NAME: /s/ Mark Smith	TITLE: CEO		
WITNESS:	Eileen Hafford	NAME: /s/ Eileen Hafford	TITLE: Assistant		
DESIGN-BUILDER: THE CHRISTMAN COMPANY					
BY:	Steven Roznowski	NAME: /s/ Steven Roznowski	TITLE: President and Chief Executive		
BY: WITNESS:	Steven Roznowski Sandy Jasinski	NAME: /s/ Steven Roznowski NAME: /s/ Sandy Jasinski	TITLE: President and Chief Executive TITLE: Exec. Assistant		